Explaining Differences in the Returns to R&D in Argentina

The Role of Contextual Factors and Complementarities

Valeria Arza Xavier Cirera Agustina Colonna Emanuel López

WORLD BANK GROUP Finance, Competitiveness and In

Finance, Competitiveness and Innovation Global Practice April 2020

Abstract

Argentina's private investment in research and development is well below that of its peers. One important reason may be low and very heterogeneous returns to research and development activities on productivity. This paper uses novel microdata to estimate the returns to research and development and understand the contextual factors that shape their heterogeneity. The paper groups these context-based factors into knowledge complementary factors (that is, factors that affect the returns via learning capabilities from external sources of knowledge) and market complementary factors (factors that act via business capabilities to appropriate the returns to research and development investments). The paper hypothesizes that the effects of contextual factors depend on firms' management capabilities and attitudes (innovative capacity), which determine firms' ability to benefit from the context. The findings suggest that the returns are indeed heterogeneous across regions and sectors, and these results depend on some context-based factors, which can boost or depress the returns to R&D. The results have important policy implications, considering the effectiveness of innovation policies, need for adapting to specific regions and sectors, and maximization of the impact of these factors on the returns to research and development.

The Policy Research Working Paper Series disseminates the findings of work in progress to encourage the exchange of ideas about development issues. An objective of the series is to get the findings out quickly, even if the presentations are less than fully polished. The papers carry the names of the authors and should be cited accordingly. The findings, interpretations, and conclusions expressed in this paper are entirely those of the authors. They do not necessarily represent the views of the International Bank for Reconstruction and Development/World Bank and its affiliated organizations, or those of the Executive Directors of the World Bank or the governments they represent.

This paper is a product of the Finance, Competitiveness and Innovation Global Practice. It is part of a larger effort by the World Bank to provide open access to its research and make a contribution to development policy discussions around the world. Policy Research Working Papers are also posted on the Web at http://www.worldbank.org/prwp. The authors may be contacted at xcirera@worldbank.org.

Explaining Differences in the Returns to R&D in Argentina: The Role of Contextual Factors and Complementarities ¹

Valeria Arza, Xavier Cirera, Agustina Colonna, and Emanuel López²

JEL: D24, L20, O32

Keywords: R&D, Argentina, Productivity, Complementary Factors.

¹ We thank the ENDEI team working at the National Directory of Scientific Information for their willingness to facilitate access to information and address our questions regarding the survey. We also thank Chad Syverson, Bernardo Diaz de Astarloa, and Cecile Thioro Niang for comments on earlier versions of the paper and Wendy Brau for excellent research assistance. This paper is a background paper to the World Bank Argentina Productivity Project.

² Valeria Arza, National Scientific and Technical Research Council (CONICET), Research Center for Transformation (CENIT), Economics and Business School, National University of San Martín (UNSAM) - Argentina

Xavier Cirera, Senior Economist, Finance, Competitiveness and Innovation, The World Bank.

Agustina Colonna, Research Center for Transformation (CENIT), Economics and Business School, National University of San Martín (UNSAM) - Argentina

Emanuel López, National Scientific and Technical Research Council (CONICET), Research Center for Transformation (CENIT), Economics and Business School, National University of San Martín (UNSAM) – Argentina.

Explaining Differences in the Returns to R&D in Argentina: The Role of Contextual Factors and Complementarities

1. Motivation

Investments in knowledge capital and other activities related to innovation are considered a key enabling factor in the catch-up process, in terms of both scaling up the technological ladder and promoting processes of reallocation and "creative destruction" (Schumpeter 1942, Griffith, Redding, and Reenen 2004). Since the early 1990s, there has been a wide consensus in the economic literature about the relation between investments in innovation, productivity, and economic growth. Evidence on this relation has been produced at the macro, sectoral, and micro levels: see, for example, Teixeira and Queirós (2016) (macro level); Strobel (2012) (sectoral level); and Crespi and Zuniga (2012) (micro level).

The creation of new knowledge that supports innovation is a cumulative process. Thus, past performance has an important influence on firms' knowledge efforts and subsequent performance. Because the past matters, various structural characteristics (such as size, age, region, and sector) can be expected to influence different patterns of investment in innovation. In addition, the effectiveness—the returns—in which investments in innovation are converted into productivity or sales growth can also vary (Fung 2004, Marín and Petralia 2018). For example, one peso invested in R&D in wine production in the Mendoza region may not enhance productivity in a similar way to one peso invested in R&D in wine productions, access to inputs, and market context are different, creating distinct opportunities in complementing private efforts in R&D.

In other words, investment in knowledge is not directly transmitted into performance (Ngai and Samaniego 2011). It depends on the particular circumstances in which investment takes place. Every context offers a different array of complementary factors that are needed to make the most of internal efforts. Whether or not several of these factors are present and the extent to which they do or do not complement one another may alter the returns to innovation.³

In addition, returns to innovation may differ in relation to firm-level heterogeneity. Acemoglu et al. (2018) use the term "innovative capacity" to describe those micro-level attributes that result in different R&D performance among firms. Cirera, Maloney, and Sarrias (2019) suggest that some of this heterogeneous innovative capacity in transforming knowledge investments into productivity gains is related to the quality of managerial practices. These firm-specific characteristics may affect the degree to which the firm is able to benefit from its context (that is, how sensitive it is to the factors previously mentioned). Hence, this paper suggests that these firm-level management attributes may interact with contextual factors, increasing the heterogeneity of R&D returns among firms.

³ The existence or lack of key contextual factors can affect the returns to R&D. What strictly speaking makes them complementary or not is whether the effect on productivity is larger when these factors and R&D are present at the same time—the final impact on productivity is larger than the sum of the individual effects.

Heterogeneity in the returns to R&D activities is of paramount importance for policy. Firms are expected to invest more in those innovation activities that yield positive returns; in other words, if the returns are low, as in the case of many developing and middle-income countries (Goñi and Maloney 2017), it is unlikely that firms will invest in innovation activities. Policies that aim to encourage investments in R&D but cannot affect these returns will fail, especially in sectors and regions with low returns. Therefore, understanding what factors explain these low returns and the heterogeneity in attaining them is critical for policy design and effectiveness.

In this paper, we examine the hypothesis that several contextual factors determine both the set of technological possibilities and the probabilities the innovation will succeed (in terms of higher productivity), as well as firms' capacity to appropriate the rewards of innovative investments in their production process. We call these factors, respectively, *knowledge complementary factors* and *market complementary factors*. We also hypothesize that the impact of these complementary factors is mediated by the firm's innovative capacity, which is manifested in specific management strategies that can maximize the returns when these complementary factors are available.

This study contributes to previous literature because it is the first to empirically analyze the relationship between contextual factors and R&D returns in Argentina with micro-level data in a comprehensive set-up.⁴ In Argentina, firms' context is frequently mentioned as an important factor incentivizing firms' defensive or short-term low-risk strategies, instead of pursuing longer-term innovative strategies that could boost productivity levels (Katz 2000, Katz and Bernat 2011, Arza 2013, Kosacoff 2000, Chudnovsky 2001, Fanelli 2002). The novelty of our approach lies in the comprehensive evaluation of this hypothesis in an econometric framework. This allows us to estimate the relationship between innovation returns and the context dimensions (which we group into "knowledge" and "market" factors), while also considering firm-specific innovative capacity. In addition, this is one of the first papers that uses data from the second wave of the Employment and Innovation Dynamics National Survey (known as ENDEI, for its Spanish acronym) (ENDEI 2014–2016). By merging ENDEI data with contextual characteristics defined at the sectoral and/or regional level,⁵ this paper also improves data availability by creating a novel and original database that could be useful for future studies.

The remaining of this paper is organized as follows. Section 2 describes the Argentinean context and its R&D and productivity performance to motivate the discussion. Section 3 describes the conceptual framework used. Section 4 describes the methodology and data. Section 5 presents results of the estimation of the production function. Section 6 concludes by offering policy recommendations, suggesting lines for future research, and discussing implications for data collection. The appendixes provide supplemental data and analysis.

2. The Argentinean Context

Although Argentina's expenditure on R&D relative to GDP or in US dollars per capita is much lower than that in developed countries, it is higher than that in regional peers, with the exception

⁴ At the macro level, the potential impact of these complementarities has been explored in, for example, Klenow and Rodriguez-Clare (2005), Maloney and Rodríguez-Clare (2007), and Goñi and Maloney (2017)—but not specifically for Argentina. These studies stress the importance (or the lack) of these complementary factors in explaining differences in the returns of R&D across countries and in the distance to the technological frontier.

⁵ This database was built using a variety of different data sources, each originally with different levels of disaggregation. Details are presented in appendix A, table A.1.

of Brazil (see figure 1).⁶ Argentina has increased the amount of resources committed to R&D over the past few years. Between 2007 and 2016, Argentina increased its total expenditures in R&D by 78 percent—measured in current US dollars in purchasing power parity (PPP) terms—and its R&D/GDP ratio by 15 percent (from 0.46 percent to 0.53 percent). These levels of growth are similar to Chile (16 percent), Mexico (16 percent), and Brazil (18 percent).



Figure 1. R&D Expenditures, Argentina and Comparators, 2007–16 (percent of GDP)

However, Argentina lags behind regional peers in private R&D expenditures (figure 2) and does not seem to be catching up. The share of private R&D expenditures in total R&D expenditures fell by 21 percent between 2007 and 2016, reaching a level of merely 24 percent of total R&D. This suggests a large preponderance of public R&D in total innovation investments in Argentina, larger than in peer countries—and potentially the result of low returns to R&D in private enterprises.

Figure 2. R&D Expenditures by Firms, Argentina and Comparators, 2007–16 (percent of total R&D)



⁶ Figures 1 to 3 contrast indicators on R&D and productivity in Argentina with a series of comparable countries. For these comparisons, we include one recent high-income country (Chile), two large Latin America economies (Brazil and Chile), and two countries belonging to the Organisation for Economic Co-operation and Development (OECD) (Canada and Chile).

In terms of labor productivity, Argentina's performance is similar to Chile's and better than Brazil's and Mexico's (figure 3), but falls within the low labor productivity growth characteristic of the last two decades in the region. Argentina's labor productivity increased by 7 percent between 2007 and 2016 but lagged behind Brazil (9 percent) and Chile (12 percent). These figures suggest that innovation investments, especially in the private sector, are not sufficient and/or effective enough to have an impact on the much-needed productivity growth. In addition, the fact that the ratio of labor productivity to R&D expenditures fell by 40 percent in Argentina but only by 30 percent in Brazil between 2007 and 2016 suggests that there might be room for improving the profitability of expenditures on R&D.



Figure 3. Labor Productivity, Argentina and Comparators, 2007–16

Source: The Conference Board Total Economy Database.

To fully benefit from increases in R&D expenditure and encourage private participation, it is important to identify possible inefficiencies in the role of knowledge in the production function, which might be deterring the contribution of R&D to labor productivity. In fact, when we compare R&D intensity and increases in labor productivity by sector, we do not always find the positive relation we would expect. Some sectors have high levels of R&D intensity but low productivity growth and vice versa (figure 4). In addition, an important share of firms manage to become innovators without committing any investment resources or by doing so very sporadically (Suárez, Lugones, and Moldovan 2008). This suggests that these firms are either taking advantage of their contextual environment—for example, through imitation, or linking to science and technology (S&T) organizations, or by employing skilled personnel who had worked previously in other innovative firms—and/or through the type of innovation that is very incremental and does not require any formal R&D efforts, but only imitation and technological adaptation or adoption.



Figure 4. Increase in Labor Productivity and R&D Intensity by sector, Argentina, 2014–16

Source: ENDEI 2014-2016.

Note: Increase in output per worker = 2014-16 increase in value added over total employment; R&D intensity = expenditure in internal R&D over sales 2014-16.

In sum, aggregate data from Argentina suggest there might be important inefficiencies in the role of knowledge in the production function, reducing private investments in R&D and constraining the impact of knowledge investments on productivity. This poses the question of what role the lack of key complementarities plays in the effectiveness of innovation investments and their returns. In other words, what is the role of knowledge complementary factors and market complementary factors in pushing upward or holding back the returns to private investment in innovation and what could the firm do to make the most of those factors?

3. Conceptual Framework

The relationship between innovation and firm performance has been largely studied, primarily using the extended knowledge production function approach pioneered by Griliches (1979). A key element in the estimation of production functions is the need to control for the endogeneity of investment (in knowledge stocks or physical capital stocks) and productivity (Parisi, Schiantarelli, and Sembenelli 2006, Doraszelski and Jaumandreu 2013, Harrigan, Reshef, and Toubal 2018, Siliverstovs 2016).

With the expansion of Community Innovation Surveys (CIS) in the European Union (EU) and later in other countries—Crépon, Duguet, and Mairesse (CDM) proposed a framework (Crepon, Duguet, and Mairesse 1998) that models the interdependent relation between investment in R&D, innovative outputs, and firms' productivity. This framework has been estimated in several contexts, including developing countries.⁷ Although this literature has reached consensus on the relevance of investments in innovation in terms of firm upgrading, there is less clarity about what factors affect the rate of return.

When trying to understand what factors affect these returns, it is very important to consider the systemic nature of innovation. In this view, innovation is considered to be the result of the interaction of many socioeconomic actors whose behaviors cannot be separated from other contextual characteristics, such as regulations, social norms, political values, and collective priorities (Lundvall 1992, Nelson 1993). This approach focused initially on innovation at the national level, through so-called National Innovation Systems, but later expanded to consider the concept of *systems of innovation* at regional levels (Cooke 2001, Cooke, Uranga, and Etxebarria 1997) and sectoral levels (Malerba 2002). The bottom line of this literature is that external knowledge boosts firms' *learning capabilities*, which, in turn, positively affect the return of private investment in innovation.

In addition, a more scattered literature analyzing macro-micro linkages on innovation also provides some insights on what factors could affect firms' returns to R&D through their capacity to convert opportunities into economic rewards (that is, how they could enhance firms' *business capabilities*). This strand of literature has examined how productivity or innovation reacts to macroeconomic uncertainty (Bernanke 1983, Bloom 2009, Acemoglu et al. 2003); competition (Aghion et al. 2005); institutional quality (Acemoglu 2003); and demand-pull factors (Adner 2002).

In this paper, based on the *knowledge expanded production function*, we bring together different contributions from these strands of literature, including qualitative insights, to analyze the role of contextual complementary factors and firms' capacity to enhance the returns of investment in innovation. We organize this body of literature into three groups: those studies that discuss contextual aspects that enhance firms' technological learning capabilities (that is, knowledge complementary factors); studies that discuss the role of contextual factors affecting firms' capacity of doing business or fully appropriating the rewards of their assets (that is, market complementary factors); and studies discussing the role of micro heterogeneity in taking advantage of contextual opportunities (that is, firms' innovative capacity).

⁷ Jefferson et al. (2006) used panel data from China to estimate the impact of R&D in terms of productivity and profitability. The paper establishes a lag structure to offset simultaneity biases and correct for endogeneity. It finds that R&D has positive effect on both profitability and productivity. Antoncic et al. (2007) used a structural equation framework on cross-sectional data for Slovenia and Romania to test a hypothesis about the positive impact of organizational support and alliances on innovativeness and, in turn, a positive impact of innovation on firms' performance (measured by growth, profitability, and wealth). They find empirical support for their hypotheses. Benavente (2006) applied an adapted version of the CDM framework using Chilean cross-section data. They find that neither R&D nor innovation results (share of sales of new products—so-called innovative sales) have an effect on productivity (measured as value added per worker). Crespi and Zuniga (2012) applied the CDM framework on micro data for six Latin American countries. They find that greater investment in R&D leads to a higher probability of having at least one process or product innovation. In addition, results show a positive impact of technological innovation on productivity (log of sales per employee) for all countries except Costa Rica. Moreover, they find that the magnitude of the results is very heterogeneous. Crespi, Tacsir, and Vargas (2016) used 2010 World Bank Enterprise Survey firm-level data to analyze 17 Latin American countries. Results show that investment in R&D per worker increases the probability that the firm will innovate, and that this translates into a strong increase in labor productivity (measured as log of sales per employee). These results are robust to five different measures of innovation: innovation of product or process, product innovation, process innovation, innovative sales (share of sales of new products), and filing for intellectual property rights. In Argentina, the CDM framework was first used by Chudnovsky, López, and Pupato (2004) and then by Arza and López (2010). Both papers show that investment in R&D boosts firms' labor productivity.

Figure 5 summarizes the key elements identified by different strands of the literature.



Figure 5. Conceptual Framework

Note: STI =science, technology, and innovation.

3.1. Knowledge Context

The first strand of literature investigates why some industries and regions are more dynamic than others. Studies generally attribute dynamism (or lack to dynamism) to the intrinsic characteristics of technological regimes, and systemic aspects of science, technology, and innovation (STI). These complementary factors are normally measured at the meso (sector/region) level and could be largely considered to be exogenous factors molding firms' patterns of innovation. We organize this literature into two blocs according to how these factors interact with firms' *learning capabilities*.

On the one hand, there is a vast literature on knowledge spillovers: several aspects have been identified at the regional or sectoral level that boost firms' learning capacity through their access to external knowledge. On the other hand, there is the literature assessing the contribution of STI policies in enhancing firms' absorptive or innovative capabilities, designed either to increase the likelihood of knowledge spillovers (such as investing in developing knowledge consortiums) or to boost firms' innovative capacities directly (such as training programs, technical assistance, and financing infrastructure).

3.1.1. Spillovers

In the systemic approach, learning by interaction plays a key role in boosting firms' learning capabilities through their access to external knowledge, embodied in other organizations or individuals. Roper and Love (2018) propose three mechanisms through which external knowledge could be privately appropriated: (i) active interaction, as in university-industry linkages or client-provider knowledge linkages, or other voluntary knowledge linkages, which could be more or less formal; (ii) active–non-interactive, as in imitation, reverse engineering, or access to codified knowledge; and/or (iii) knowledge externalities, as in social networks,

which in turn positively affect the return of private investment in innovation. In this work, we refer to positive *spillovers* on firms' productivity that are originated by knowledge-related activities carried out by other organizations or individuals in their immediate context. These spillovers can be sectoral or regional/spatial.

a) Sectoral context

The literature on *technological opportunities* (TO) argues that the sectoral differences in the returns to R&D are explained by variations in TO. These are defined as the set of technological options that determine the distribution of the values of the production function or product attributes that can be achieved via R&D (Klevorick et al. (1995)). Although this concept is fundamentally technologically driven, the preoccupation with decreasing returns of investment in R&D pushes scholars to analyze systemic sources claimed to be able to renew the pool of TO. These are explained either by technological trajectories—as in the traditional literature arguing that technological specificities drive different patterns of innovation across sectors (Malerba and Orsenigo 1995, Pavitt 1984)⁸—or related to spillovers created either by –STI policy (particularly in supporting universities and research institutions that produce market-relevant scientific knowledge) or by advancement in the industry. This has been assessed empirically in Klevorick et al. (1995) and later contributions, including Marín and Petralia (2018); Fung (2004); and Kafouros and Buckley (2008).⁹

b) Regional/spatial context

Several contributions concerning the systemic approach come from regional innovation studies analyzing how socioeconomic factors present in the territory where the firm produces (such as the institutional capacity, the supply of human capital, and public support for the generation of knowledge) affect both the level and the effectiveness of their investment in R&D (Crescenzi and Rodríguez-Pose 2013). This literature is motivated by the stylized fact that innovation is not randomly distributed in the territory but tends to concentrate geographically (Audretsch 1998). As far back as 1890, Marshall (1890) identified three reasons that explain geographical concentration: availability of skilled workers; availability of specific inputs, such as natural resources; and technological spillovers. This rationale gave birth to the new economic geography and the study of economics of agglomeration, led by Krugman (1991). According to his studies, "geographic concentration of production is clear evidence of the pervasive influence of some kind of increasing returns" (page 6).

3.1.2. Science, Technology, and Innovation Policies

Policy instruments are designed to strengthen several aspects needed to increase the rate of return of R&D, such as a supportive scientific system; providing infrastructure for interaction

⁸ Pavitt (1984) classifies industries according to the source of technology (such as inside/outside the firm, government/private-financed); users' needs (such as price/performance); and methods used to appropriate benefits from innovation (such as secrecy, patents, time lags, and unique knowledge). Similarly, Malerba and Orsenigo (1995) explore the dynamics of technological change and define two different groups, labeled technological regimes, that are characterized by a specific combination of conditions of technological opportunity, appropriability of innovation, cumulativeness, and properties of the knowledge base defined at the sectoral level. ⁹ Another important related strand of literature summarized in Griliches (1994) and more recently in Bloom et al. (2017) focuses on the productivity of new ideas. This literature is more macro but has sector-specific implications because in some sectors ideas "are harder to get" and this will affect their returns to R&D investments to develop them.

and learning; supporting adequate financing schemes to tackle technical challenges; addressing market failures and/or guiding market signals to reflect and internalize the social returns to innovation; building an institutional (regulatory) environment to promote certain strategic innovation paths; and/or guiding the generation of opportunities and promoting interactions between the actors sharing that vision of development (OECD 1998).

Innovation policy also has a key role in creating a collective vision of what should be done and how to enhance innovation performance (Metcalfe and Ramlogan 2008, Soete, Verspagen, and Ter Weel 2010, Kline and Rosenberg 1986). Whether all or some of these aspects do or not exist affects the effectiveness of any private effort in innovation. STI policies, therefore, are a *complementary factor*, mostly affecting firms' *learning capabilities* but also affecting firms' *business opportunities*. For example, regulatory aspects of STI instruments such as intellectual property schemes, or specific instruments such as public procurement, grants, or other monetary incentives, could improve the rewards to innovation, and therefore STI policy, thus also constituting an important *market* complementary factor (included in the Regulation segment in figure 5).

3.2. Market Context

While traditionally the National Innovation System literature has emphasized the systemic nature of innovation and the importance of key complementary factors and institutions in defining innovation outcomes, it has often lacked clarity on what context factors matter and are complementary to R&D. Maloney (2017) proposes where to set the boundaries of the NIS while emphasizing the often underacknowledged role of barriers to accumulation of factors of production. In the context of a knowledge production function, knowledge inputs are often affected by barriers that deter the accumulation of other factors: that is, barriers to capital investment due to uncertainty or financial market imperfections are also barriers to invest in knowledge capital. A key insight is that in addition to the knowledge context, key aspects of the traditional market context will also have an impact on the returns to investment in innovation.

3.2.1. Competition

Market competition has been considered an important aspect promoting or deterring innovative behavior. There is general consensus about the simultaneous determination of market competition and innovation. The concepts of creative destruction and creative accumulation in Schumpeterian approaches are evidence of an awareness of this simultaneity. In a regime characterized by creative destruction, entrepreneurs are driven by fear of others innovating first. Therefore, the innovative base is continuously being enlarged by the entry of new innovators, increasing market competition. In a creative accumulation regime, on the other hand, it is monopoly power that encourages innovation, which in turn will be rewarded by monopoly rents, increasing market concentration.¹⁰ Thus, market competition may affect both incentives to invest and opportunities to appropriate rewards from such investment, which makes it a central market complementary factor.

¹⁰ Many empirical studies have attempted to find economic mechanisms to explain the highly concentrated market structure of a highly R&D-intensive sector; since Scherer (1967), the relation between market competition has been modeled in an inverted-U shape. However, because these variables are related in both directions, the methodological challenge in empirical studies has been how to account for such endogeneity: see Davies and Lyons (1996), Sutton (1998), Aghion et al. (2005).

3.2.2. Demand

A similar argument could be made in relation to market demand. There is a long tradition in innovation studies highlighting the role of demand in pulling technological progress, often based on Kaldor's ideas about the importance of demand in encouraging investment and the virtuous aspects of high income demand elasticities affecting rates of returns in investing in certain sectors (Dixon and Thirlwall 1975, Kaldor 1966, McCombie and Thirlwall 1995, Schmookler 1962). Demand-pull factors have since been considered key aspects of guiding investment in innovation and explaining heterogeneity in return rates. In turn, lack of demand or uncertainty about future demand are considered deterrents to firms' decisions to innovate (see, for example, García-Quevedo, Pellegrino, and Savona (2016)).

3.2.3. Uncertainty

The issue of economic uncertainty in general requires specific consideration as a condition that could affect market and learning opportunities and therefore firms' attainment and appropriation of innovation rewards. A large array of studies analyzes the negative impact of uncertainty on investment due to the existence of irreversibility.¹¹ A firm has more flexibility to decide between inputs, technologies, and organizational set-ups before it makes the decision to invest in a particular machinery or to initiate a specific R&D project. Then, if relevant variables (such as interest rates, asset prices, exchange rates, input prices, and labor costs) are uncertain, the firm might decide to postpone or cancel its investment decisions, either because it cannot foresee future returns on those investment projects in such an uncertain context or because the expected rate of return that compensates for the increased risk is unachievable. The acquisition of knowledge is not automatic, and learning requires foresight over the medium term. Thus, when a context is highly uncertain and macroeconomic changes cannot be fully anticipated, investment behavior will be limited and contemporaneous rewards on past investments will be reduced, which justifies our consideration of uncertainty as a *market complementary factor*.¹²

3.2.4. Regulations and Business Environment

Several policy regimes, such as labor, tax, trade, foreign direct investment, and/or competition policy, affect knowledge investment decisions and their impact. For example, the ability to hire engineers and move across tasks in the plant, import machinery, or hire foreign managers affect the effectiveness of R&D. More generally, the regulatory and business environment affects the incentives to invest in innovation activities and the returns to these investments in general. This therefore also makes it an important *market complementary* factor.

¹¹ The papers by Caballero and Pindyck (1996) and Pindyck and Solimano (1993) show that the threshold of the marginal return on capital that triggers investment increases with the volatility of the marginal return, and therefore investment decreases with volatility. Caballero and Pindyck focus on US manufacturing industries, while Pindyck and Solimano's contribution is a cross-country study. Indeed, they find that the impact is larger for developing countries (see Pindyck and Solimano (1993): 33).

¹² Given that a less uncertain environment also favors learning, as discussed, innovative capacity could also be considered a knowledge complementary factor.

3.3. Innovative Capacity

Firms' ability to transform knowledge investments into productivity gains also varies. The bottom line of this strand of literature is that there are differences in firms' innovative capacity that translate into heterogeneity in performance.¹³

This heterogeneity is also likely to mediate the impact of the context and potential complementary factors already discussed. STI support for example, is likely to be effective for firms with greater innovative capacity. Similarly, more innovative firms are more likely to take advantage of a good supply of engineers and better demand conditions or locational spillovers.

In sum, our conceptual framework anticipates that there are knowledge and market factors stemming from the firms' context that complement private investments in knowledge capital. In addition, some firms are better prepared to take advantage of those factors. All in all, the existence of context-based factors and firms' innovative capacity to make the most of them could explain significant heterogeneity in returns. We therefore propose two hypotheses regarding the role of context-based factors on returns to private investment in innovation (against the null hypothesis that context-based factors do not matter for returns to innovation).

H1: Context-based factors directly affect the returns to in-house investment in innovation.

H2. Context-based factors affect the returns to in-house investment in innovation when mediated by firms' innovative capacity.

These hypotheses are summarized in figure 6. Our goal in the sections that follow is to empirically identify the key factors that may, according to H1 and H2, affect R&D returns in Argentina, with the aim of improving policy design.

Figure 6. Main Hypotheses under Study



¹³ Acemoglu et al. (2018) emphasize that some of the distortions can be associated with industrial policy. Public support to firms that have low quality of innovation—firms with little ability to convert innovation investments into productivity— will result in lower productivity gains and as a result lower aggregate returns.

4. Methodology

4.1. Data Sources

Micro Data

The main database for our analysis is the second Employment and Innovation Dynamics National Survey (ENDEI 2). The use of this data set is completely novel. Access was granted to us by the National Directorate of Scientific Information (Dirección Nacional de Información Científica, DNIC) and the Under-secretariat of Studies and Prospective (Subsecretaría de Estudios y Prospectiva), specifically for the needs of this project.¹⁴ This survey covers the 2014–16 period and was carried out jointly by the Labor and Employment Secretariat (Secretaría de Trabajo y Empleo) and the Science, Technology and Productive Innovation Secretariat¹⁵ (Secretaría de Ciencia, Tecnología e Innovación Productiva). The sample was drawn to be representative of manufacturing firms with at least 10 employees, in terms of size (small, medium, and large firms), region (five geographical areas), and sector (mostly at the 2-digit ISIC level).¹⁶ The sample includes 3,945 firms. For some exercises, we used the previous ENDEI (hereafter ENDEI 1), comprising 3,691 firms, which covers the period 2010–12 and which is representative at the size and sectoral level. We could not use panel data because both waves were not matched at the firm level.¹⁷

The ENDEI has two structured questionnaires, one self-administered and one that requires a face-to-face interview. The former contains questions that require inputs from different areas of the firm: income; expenses (such as wages and salaries, intermediate consumption, and investment in fixed assets);¹⁸ employment (according to hierarchies and qualification); and remuneration and spending in innovation activities (such as R&D, consultancy, and acquisition of machinery and equipment). The latter contains mainly qualitative information on several issues regarding innovation and employment dynamics: organizational capability and business

¹⁴ We were allowed access to this information with an agreement in 2019 between the World Bank and the Ministry of Production and Labor (Ministerio de Producción y Trabajo). The database was completely anonymized and prepared specifically for the needs of this paper. In order to respect the confidentiality agreement with the firms surveyed, we worked with a computer in the offices of the DNIC, under constant supervision of the team.

¹⁵ The former Ministry of Science and Technology, which was transformed into a Secretariat in September 2018. At the end of 2019, the institution returned to a Ministry structure again.

¹⁶ ISIC, the International Standard Industrial Classification of All Economic Activities, is the international reference classification of productive activities. The following sectors are included in this study: Food, beverages and tobacco; Chemicals and petrochemicals; Pharmaceutical; Basic metals; Motor vehicles, ships and other transport equipment; Paper and publishing; Rubber and plastic; Machinery and equipment; Textiles and wearing apparel; Electrical machinery and apparatus, TV and radio equipment; Wood and products of wood; Leather and footwear; and Other industries. For some sectors of special interest, information was disaggregated at 4 digits (Food and beverages; Chemicals; Machinery and equipment and Motor vehicles). The five regions were: Patagonia (including the provinces of Chubut, Neuquén, Rio Negro, Santa Cruz and Tierra del Fuego); Cuyo (including the provinces of Mendoza, San Juan and San Luis); the Northern region (including Chaco, Corrientes, Formosa, Misiones, Catamarca Jujuy, La Rioja, Salta, Santiago del Estero and Tucuman); Pampeana (including Buenos Aires, Cordoba, Entre Rios, Santa Fe, and La Pampa); and the region of the Capital city and suburbs.

¹⁷ However, it is worth highlighting that sampling methods changed between both waves. In this respect, only ENDEI 2 is relevant for our exercise of assessing the role of regional/sectoral contextual factors because ENDEI 1 is not representative at the regional level.

¹⁸ Unfortunately, neither information for energy consumption nor investment in physical assets was correctly measured in ENDEI 2. Although those questions were included in the questionnaire, data were not made available because they include severe measurement errors. We needed to develop strategies to overcome these data constraints to estimate capital stock in the production functions, as discussed later in the paper.

strategy; innovation activities; profile of human resources dedicated to innovation activities; results of the innovation efforts; sources of information and innovation objectives; sources of finance for innovation activities; obstacles to innovation; linkages; employment management capabilities and training policy; organization of labor; and knowledge management capabilities.

Sectoral and Regional Data

Table A.1 of appendix A presents all indicators considered in the analysis as proxies of contextbased complementary factors. They are organized according to our conceptual framework, presented in figure 5. These indicators were built from various data sources listed in the last column of the table.

4.2 Empirical Strategy

4.2.1 Introducing Innovation Complementarities in the Production Function

In order to estimate the average returns to innovation across all firms in our sample, we first estimate the knowledge production function specified in equation [1]. Following Griliches (1979), this baseline equation is a production function in per worker units: that is, extended by knowledge. The dependent variable is the aggregate value¹⁹ of firm *i* at year *t*. Inputs are labor (L), capital stock (C) and knowledge stock (I). We measure all variables in natural logarithms, assuming a log linear relationship between inputs and aggregate value. Firm and time fixed effects are included in order to control for firm and time invariant unobserved heterogeneity.

$$\frac{Y_{it}}{L_{it}} = \alpha + \beta L_{it} + \gamma \frac{C_{it}}{L_{it}} + \delta \frac{I_{it}}{L_{it}} + u_i + \tau_t + v_{it} \quad . \quad [1]$$

Given that we aim to measure the effect of context-based complementary factors on innovation returns, the baseline model expressed in equation [1] is expanded to include interactions between knowledge (K_{rst}) and market (M_{rst}) factors—defined at the sectoral (s) and regional (r) level— and investment in innovation. This allows us to recover the influence of the different factors over the innovation returns coefficient (equation [2]).

$$\frac{Y_{irst}}{L_{irst}} = \alpha + \beta L_{irst} + \gamma \frac{C_{irst}}{L_{irst}} + \delta_0 \frac{I_{irst}}{L_{irst}} + \delta_1 M_{rst} + \delta_2 K_{rst} + \delta_3 M_{rst} \frac{I_{irst}}{L_{irst}} + \delta_4 K_{rst} \frac{I_{irst}}{L_{irst}} + u_i + \tau_t + e_{irst}.$$
[2]

Therefore, δ_0 , δ_3 , and δ_4 are our main coefficients of interest because they estimate the marginal effect of market and knowledge complementary factors on innovation returns. Standard panel data procedures are used to produce estimates for coefficients.

4.2.2 Endogeneity of Investment in Innovation

The likely presence of omitted variables and simultaneity issues raises a challenge for the precise identification of the coefficients of the knowledge production function. In this section, we focus on the endogeneity of the investments in innovation variable, given that its returns over productivity are our main concern. We are aware that capital and labor variables are

¹⁹ Results are robust when we use sales as the dependent variable and include expenditure on intermediate goods in the regression.

probably also endogenous, but we do not take explicit measures with them because their returns lie beyond our research questions. Time and individual fixed effects, however, tackle issues of omitted variables for all independent variables, decreasing bias of all estimated coefficients.

Given that firms construct their knowledge stock by investing in innovation activities, these decisions are very likely endogenous to firms' productivity, and exogeneity of regressors cannot be assumed. Unobservable omitted variables such as know-how the firms' workers possess, or the quality of managerial capabilities could affect both firms' decisions regarding innovation and firms' productivity. In addition, while larger knowledge stocks may increase firm productivity, more productive firms are more likely to be exposed to and aware of innovation opportunities. Hence, these firms may be more prone to investing in innovation and increasing their knowledge stock than less productive firms, causing reverse causality issues.

We tackle time-invariant omitted variables—both observable and unobservable—by exploiting the panel structure of our data with the inclusion of firm and time fixed effects, as has been done in other papers using ENDEI survey databases (see, for example, Brambilla and Tortarolo (2018)). We also explore an instrumental variable (IV) approach to estimate the returns of R&D over productivity. Claiming exogeneity of variables coming from our own ENDEI database is questionable and the restrictions to merge other potential instruments from external sources at the firm level due to ENDEI's confidentiality issues limits our analysis. However, it is important to try to minimize potential biases due to the endogeneity. Next, we describe the IV strategy and present the best IV candidates. As we shall see, the instruments proposed are statistically strong and exogenous, but the results of the IV estimation differ sharply from those of our panel data estimation, which raises questions about the accuracy of the point estimates.

Our first IV candidate is a dummy variable on firm's *knowledge* about the existence of public support programs for innovation activities offered by the Ministry of Science and Technology (Ministerio de Ciencia y Tecnología). This variable has been implemented in previous empirical studies that use information from innovation surveys in Argentina (see, for example, Brambilla and Tortarolo (2018), de Elejalde, Giuliodori, and Stucchi (2011). The condition of relevance is expected to be fulfilled because firms that know about public programs are candidates to use it, and therefore have a higher investment in innovation activities. The exclusion condition is justified considering that this dummy variable measures *knowledge* about the program rather than firms' participation. The latter can be correlated with other variables that may affect firm's productivity, given that the provision of public funds is sometimes based on firm performance. On the other hand, knowledge of the program greatly depends on public and private publicity for the program, which we can assume is random and uncorrelated with other firm characteristics.

Our second IV candidate exploits a pseudo-panel built by merging the two waves of the ENDEI survey. Although we cannot match firms across waves, we proceeded to match them at the sector-region-size level.²⁰ An essential reason for choosing this IV candidate is that there was a change in government in Argentina in December 2015.During the ENDEI 1 period (2010–12) and the first two years of the ENDEI 2 period (2014–15), the leading political party differed

²⁰ This is possible given that samples were constructed to be representative of the Argentinean manufacturing sectors (by industry-size). However, because ENDEI 1 was not constructed to be representative at the regional level, we could not divide the sample into the same regions as ENDEI 2. We therefore made an ad hoc split of the sample considering that cases were relatively balanced: "Region" is taken as a dichotomous variable signaling whether the firm belongs to the Gran Buenos Aires region (comprising Ciudad de Buenos Aires and the main adjacent districts in Buenos Aires Province) or to the rest of the country.

from the one during the last year covered by the ENDEI 2 survey. While regulations and restrictions on trade policies were prevalent during the ENDEI 1 reference period— especially when the survey was conducted (2013)—they fell off sharply during the last year covered by the ENDEI 2 survey (2016) and dropped even more when it was conducted (2017–18).

This IV strategy rests on the idea that regulations and restrictions on trade policies imposed by the government affect the perceived obstacles to innovation in a heterogenous way for firms of different sectors, regions, and sizes, which in turn may limit the resources they dedicate to innovation activities. In particular, we hypothesize that import barriers on key goods needed to carry out innovation activities are sensitive to regulations and trade policies. We would then expect that changes in trade restrictions in 2016 decreased perceived obstacles to imports of goods for innovation activities, and consequently fostered investments in innovation between 2014 and 2016.

Hence, we proxy a firm's restrictions on imports in 2014 through the sector-region-size *proportion of firms* claiming to suffer from import barriers during the ENDEI 1^{21} period; we assume that firms belonging to the same group are similarly affected by these barriers. Hence, the variability in our instrument comes from differences in the intensity of perceived obstacles to imports across sector-region-size groups of firms. We then observe how these perceived obstacles relate to the *change* in innovation investments between 2014 and 2016, given that between these two years many import restrictions were lifted. We expect a positive correlation between the perception of obstacles in the past and the change in innovation investment, given that firms that were more restricted are the ones that increased their investment in innovation activities to a greater extent once these restrictions were removed. The conditional exogeneity assumption is plausible, given that the effect of import restrictions on goods that *are key for innovation* affects productivity levels precisely through their effects on the firm's decision to conduct or increase its innovation activities, once these barriers are alleviated. A graphical explanation of the logic of this instrument is presented in figure B.1 in appendix B.

It is important to highlight that the constructed IVs are cross-sectional and are used to instrument the *difference* in innovation investment between 2014 and 2016 (we are left with only one observation per firm). As a result, the database loses its panel data structure. First differences, however, eliminate the effect of time-invariant omitted covariates.

4.3 Measurement Issues

4.3.1. Measuring the Capital Stock

To estimate equation [1], we needed to produce estimations for the firms' capital stock because data on this variable are not collected in innovation surveys. Due to serious measurement errors, ENDEI 2 does not provide information about two variables that could have been used as proxy of physical capital stock (which were provided in ENDEI 1):

1. Consumption of energy, gas, and fuel. This variable could have been used as a proxy under the argument that industrial capital stock should increase at a similar rate as the consumption of energy used to operate such equipment (Frank 1959).

²¹ We assume that restrictions present during the ENDEI1 period (2010–12) can represent restrictions present in 2014, given that both the political administration and the trade regulations were the same during both periods.

2. Investment in machinery and equipment. This is a flow variable accounting for gross fixed capital formation. It can help build the capital stock through the permanent inventory method.

Having tested that energy consumption worked well as proxy of capital stocks in production function using ENDEI 1, and given the absence of alternatives in ENDEI 2, we estimate energy consumption using predictions based on variables aggregated according to firm sector and size from ENDEI 1 to be applied to ENDEI 2 (see appendix C for a description).²²

4.3.2 Methodologies to Measure Knowledge Stock

Since Griliches' seminal study in 1967, there has been an intense academic debate about how to measure the knowledge stock using investment in innovation.²³ First, there is a problem of aggregation of knowledge, because various types of knowledge are useful to the production process. Some types are embodied in people or even in the organization and some can be acquired from external parties or by doing research in-house. The literature has normally used in-house investment in R&D as a shortcut, mostly because it is easier to measure than, for instance, organizational knowledge, and it could be argued it may be complementary to any other source of knowledge. R&D is then, at best, a representative input among others in the process of knowledge capital formation.

Second, there is some debate about what the relevant lag structure involved in estimation of current knowledge stock is: that is, it might take more than one year to complete a *research* project. When that project is completed (and if it is successful), the *development* part of the project might take even longer. Moreover, knowledge formation is a cumulative process: that is, what has been done in the past is relevant for building new knowledge in the present.

But what is the appropriate lag? The rate of depreciation may be high: Mansfield (1972) estimated an R&D lags structure, finding peaks three to five years after the investment and then declining rapidly. Little remains "private" 10 years after investment. Unfortunately, we do not have a long enough database of firms' innovation efforts to include lags in our estimation of the knowledge stock. We then need to rely on contemporaneous investment as a proxy of knowledge stock. This choice is justified by the explanation that past decisions have a great influence on both current decisions and on performance resulting from those decisions. Part of the outcome of investing in R&D is improved capacity to innovate in the future: that is, investments in R&D build the knowledge stock useful for future production. Thus, when firms decide to invest in innovation, they anticipate such efforts need to be sustained in the near future. Investment in R&D is therefore fairly sticky (Dosi 1988), and is not expected to be subject to as severe changes at the micro level as a proportion of sales or value added. Hence, what the firm invests today, relative to other firms, could be a relevant proxy of what the firm has invested in the past in relative terms.

One final key methodological element relates to the measure of all relevant types of knowledge

²² We tested different alternatives to measure capital stock. Among them we also used the one proposed by Galiani, Gomez, and Scattolo (2019), also part of this project. All results are strongly robust. Our measure allows more variability of capital stock across sectors without information loss (no missing values are created with our procedure). Given that sectoral and regional sample representation is key to answer our research question, we prefer to use our proposed proxy of capital stock.

²³ Griliches (1967) estimated that firm's R&D and firm's productivity are connected in a bell-shaped lag structure. Since then, several strategies have been followed, typically using R&D and its lags.

besides R&D. In this paper, we also consider investment in design and industrial engineering. The reason for this is twofold. First, in several types of manufacturing (such as making wearing apparel), knowledge is incorporated mainly in the production process design. Because potential complementary factors are measured at the sectoral level, we need to be sure we incorporate the sectoral specificities of knowledge stock formation. Moreover, industrial engineering is particularly relevant for reverse engineering and technology adaptation, which is the typical first stage in innovation learning (Katz 1982). Second, design and engineering implies in-house efforts that often cannot be distinguished from R&D (especially development) efforts (Cox 1990). In many firms, these types of knowledge activities are performed mainly by the same human resources and are very difficult to disentangle one from another. Indeed, all guidelines for innovation surveys globally explicitly discuss the difficulties of differentiating both types of knowledge. This is also the case for ENDEL.²⁴ Thus, for conceptual and methodological reasons, it is better to add design and engineering to R&D (RD&D, research design and development) in our measurements of firms' knowledge stock formation. Figure 7 presents average levels of RD&D per worker per productivity percentile, clearly showing a positive relationship, as expected with a good measure of knowledge investments.

Figure 7. Average Investment in RD&D per Worker per Each Percentile of Labor Productivity



Note: RD&D = research design and development.

²⁴ We cite the ENDEI questionnaire: "Industrial Design and Engineering Activities: they are those activities carried out within the firm: technical functions for production and distribution not included in R&D, drawings and graphics for establishing procedures, technical specifications and operational characteristics; installation of machinery; industrial engineering; and production start-up. These activities can be difficult to differentiate from R&D activities; for this it can be useful to check if it is a new knowledge or a technical solution. If the activity is framed in the resolution of a technical problem, it will be considered within the Engineering and Industrial Design activities. It should include the annual salary of the staff devoted to these activities according to the time dedicated."

5. Results

5.1. Production Function

Table 1 shows results for the baseline estimation of equation [1]. The dependent variable is value added per worker. Production factors include labor, physical capital, and knowledge capital (proxied by energy consumption and RD&D, respectively, as previously explained) and rescaled to per worker units. All variables are expressed in natural logarithms.

In addition to the baseline model, we include alternative specifications in the table to check for (i) the effectiveness and significance of estimation methods in dealing with potential endogeneity issues, and (ii) the robustness of estimated coefficients, mainly for the knowledge stock proxy, which is the main focus of the study. We analyze the stability of coefficients' running pooled ordinary least squares (OLS) (column 1), panel (column 2), and second-difference (2014–16, column 4)²⁵ models using our preferred set of explanatory variables expressed as per worker ratios. We also conduct panel regressions using variables in levels instead of the per worker units (column 3); and pooled (column 5) and panel (column 6) regressions using R&D instead of RD&D as proxies of knowledge stock.

Our preferred specification corresponds to column 2, where the coefficients for labor, capital, and knowledge factors are statistically significant and show the expected signs: a 10 percent increase in RD&D per worker increases productivity by 0.09 percent, and a 10 percent increase in energy consumption per worker increases productivity by 3 percent. The coefficient for labor is significant and negative, which implies decreasing returns to scale.²⁶

Results are fairly robust for different specifications. RD&D coefficients are always positive and significant, and the coefficients' size is relatively stable in panel data estimations (columns 2, 3, and 4). For pooled estimations, in contrast, the coefficient is much larger (1.7 times larger than fixed effects, FE) (column 1 against column 2), which may suggest that the effect of omitted variables on RD&D and productivity goes in the same direction. This could be the case, for example, of managerial capabilities or entrepreneurial quality, which affect both RD&D and productivity positively, in terms of better ability to manage R&D projects and production processes in general.²⁷

The coefficient for our proxy of capital stock is very robust. Labor seems to be more correlated with omitted variables, given that the coefficient is positive for the pooled estimation, but otherwise negative and consistent across panel specifications.

 $^{^{25}}$ For this specification, we took differences of all variables between 2014 and 2016 as in our IV estimation. Hence, this column allows us to compare results with and without instrumenting RD&D in the production function.

²⁶ In all estimations of table 1 except column 3, all variables are in units per worker (divided by *L*). Because our model is linear in logarithms, we can assume a Cobb-Douglas specification: $Y = AL^{\theta}C^{\gamma}I^{\delta}$, and dividing by *L*: $\frac{Y}{L} = \left(\frac{A}{L}\right)L^{\theta}L^{\gamma}L^{\delta}\left(\frac{C}{L}\right)^{\gamma}\left(\frac{I}{L}\right)^{\delta} = AL^{\beta}\left(\frac{C}{L}\right)^{\gamma}\left(\frac{I}{L}\right)^{\delta}$ with $\beta = \theta + \gamma + \delta - 1$. Hence a negative coefficient for *L* implies that $\gamma + \delta + \theta < 1$ (that is, decreasing returns to scale).

²⁷ When the knowledge stock is approximated by R&D (column 6), we also find the significant and positive effect on productivity. However, when comparing results for R&D with and without FE (columns 5 and 6), the bias is negative, which means that if it is caused by omitted variables, they are correlated in opposite directions to R&D and to productivity. This may suggest that there are omitted innovative efforts that work as substitutes for R&D (but affect productivity positively). This may be the case for design and engineering, given that, as noted, it is very difficult for respondents to empirically discriminate between them.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Variables	VA per worker	VA per worker	VÁ	Changes VA per	VA per worker	VA per worker	GMM estimates
				worker			
Canital	0 32***	0 30***	0 33***	0 37***	0 32***	0 30***	0 31***
Capital	(0.032)	(0.032)	(0.038)	(0.041)	(0.032)	(0.032)	(0.042)
RD&D	0.015***	0.0088**	0.0083**	0.0067*	(0.052)	(0.052)	0.097**
ill ab	(0.003)	(0.0035)	(0.0035)	(0.0037)			(0.048)
Labor	0.062**	-0.26***	0.30***	-0.027	0.068**	-0.20***	-0.014
	(0.028)	(0.045)	(0.11)	(0.048)	(0.028)	(0.048)	(0.048)
R&D		()		()	0.017***	0.074***	
					(0.003)	(0.019)	
Dummy 2015	0.17***	0.18***	0.18***		0.18***	0.18***	
	(0.014)	(0.015)	(0.016)		(0.013)	(0.015)	
Dummy 2016	0.35***	0.35***	0.33***		0.34***	0.34***	
	(0.021)	(0.023)	(0.026)		(0.021)	(0.023)	
Missing dummy ^a	0.061*	0.090*	0.080	0.048**	0.068*	0.85***	0.066**
	(0.033)	(0.048)	(0.048)	(0.024)	(0.037)	(0.23)	(0.029)
Constant	9.26***	10.6***	10.7***	0.33***	9.27***	10.4***	0.30***
	(0.29)	(0.38)	(0.40)	(0.027)	(0.28)	(0.38)	(0.028)
Observations	9.246	9.246	9.254	2.949	9.246	9.246	2.949
R-squared	0.159	0.390	0.397	0.055	0.158	0.393	_,, .,
Number of ID	3,489	3,489	3,490		3,489	3,489	
Firm FE	No	Yes	Yes	No	No	Yes	No

Table 1. Results for Baseline Estimation of Equation [1]

Note: FE = fixed effects; GMM = generalized method of moments; R&D = research and development; RD&D = research design and development; VA = value added.

Standard errors are clustered at the sector-size level.

*** p<0.01, ** p<0.05, * p<0.1.

a. We replaced missing values of investment in R&D and Industrial Design with zero in order to avoid losing observations. We control for this underestimation of our stock of knowledge variable by including a dummy that indicates whether the firm reported missing values in any of these categories or not. The estimated coefficient has the expected sign and is almost always statistically significant.

5.2.Instrumentation of Knowledge in the Production Function

Our procedure consists of using generalized method of moments (GMM) to re-estimate the equation presented in column 4 in table 1. As mentioned, we use as instruments both the dummy indicating knowledge of public support for innovation and the sector-region-size intensity of past perceived barriers to import key goods for innovation. IV results are presented in table 2. Column 1 presents the first-stage results and column 2 presents the GMM results (which repeat the specification in column 7, in table 1). First-stage results reveal the expected signs and significance for both instruments.

The test for weak instruments (Montiel-Pflueger test) (Olea and Pflueger 2013), which considers the clustered error structure of the models, gives evidence of rejecting weak instruments hypotheses.²⁸ In addition, because our IV model is overidentified—that is, the number of instruments exceeds the number of endogenous regressors—we can test whether instruments are uncorrelated with the error term by means of a Sargan (1958) test. In this procedure, a significant test statistic represents either an invalid instrument or an incorrectly specified structural equation. Table 2 shows that the statistic is not significant, supporting the validity of our instruments.

	(1)	(2)
Variables	First Stage	GMM estimates
Prop. of firms affected by	0.89***	
barriers to imports	(0.27)	
Knowledge of financial programs	0.22**	
	(0.094)	
RD&D		0.097**
		(0.048)
Capital	0.13	0.31***
	(0.11)	(0.042)
Labor	-0.21**	-0.014
	(0.089)	(0.048)
Missing dummy	-0.27*	0.066**
	(0.15)	(0.029)
Constant	0.0049	0.30***
	(0.11)	(0.028)
	2 0 4 0	2 0 4 0
Observations	2,949	2,949
R-squared	0.010	
Overidentif restric test		
Sargan statistic	1.79	
Pvalue	0.1809	
Weak IV test		
M-P Statistic	8.825	
tau=5% Critical Value	8.48	

Table 2. Instrumental Variables Estimation

Note: Standard errors are clustered at the sector-size level. For weak IV test, tau = % of worst case bias. RD&D = research design and development.

²⁸ The test rejects the null hypothesis that the Nagar bias of the second-stage coefficient of RD&D exceeds 5% of the "worst case bias": that is, the case in which instruments are completely uninformative and first and second stage errors are perfectly correlated.

Second-stage results show that the physical capital proxy and labor coefficients are fairly robust. The capital coefficient is close to 0.3 and labor coefficient is insignificant, as in the OLS estimation using variables in differences (table 1, column 4). The RD&D coefficient shows a positive and significant impact in labor productivity, but its magnitude is almost 15 times larger than the specification without instrumenting (table 1, column 4), which calls into question the accuracy of the instrumentation procedure.

5.3 Measuring Different Returns to Investment in Knowledge Stocks by Sector and Region

Given that the main focus of this analysis is to explain differences in RD&D returns across sectors and regions, equation [1] is estimated by sector using the specification in column 2, table 1. Coefficients for RD&D are presented in figure 8, showing significant heterogeneity and confirming that returns to RD&D are largely sector-specific.²⁹ The returns, proxied by estimates of RD&D elasticity on labor productivity, appear to be particularly large for wearing apparel (18), machinery for agriculture (2921), and pharmaceutical products (2423), although the latter coefficient is not significant.



Figure 8. Estimated Coefficients for Sectoral Returns to RD&D

Note: For a list of economic sectors, see appendix E. ISIC = International Standard Industrial Classification; RD&D = research design and development.

²⁹ F test statistic for Ho: coefficients are the same across sectors, rejects at 1% level.

Figure 9 presents the returns to RD&D when calculated separately by regions. Except for returns in the Patagonia region,³⁰ they are less heterogeneous³¹ (we suspect this may be a result of data being too aggregated at the regional level). Differences in coefficients are also statistically significant across sector/region groups.



Figure 9. Estimated Coefficients for Regional Returns to RD&D

Interestingly, the variance across sector/region groups for RD&D coefficients is much greater than the one for the coefficients for the other variables in our production function. Specifically, the coefficient of variation for RD&D coefficients across sectors is between 2.25 and 2.5 times greater than that of the coefficients for labor and capital. This seems to indicate that sectoral-and regional- specific factors have a greater effect on RD&D returns than on the other variables, leading to greater dispersion of this estimator. This raises the question of which factors are relevant to explain the larger heterogeneity in RD&D returns across sectors and regions.

This large heterogeneity across sectors and regions also appears to exist in differences in market and knowledge complementary factors. Figures B.2 and B.3 of appendix B show the variation of some these variables. For instance, inhabitants in the northern region of the country lag in economic and educational levels. Looking at sectors, the pharmaceutical products sector (2433) has larger firms or a greater supply of academic publications that are relevant to the industry. This heterogeneity in some complementary factors may explain the estimated heterogeneity in returns to RD&D.

Figure B.4 shows the relationship between certain of those market and knowledge contextbased characteristics and returns to innovation per sector. Four quadrants were built divided by the median value of contextual variables and RD&D returns as shown in Figure 8. The color

³⁰ The Patagonia region is a special case in terms of *labor* productivity because it is specialized in capital- and natural resources–intensive industries and receives strong fiscal and economic support from the state for certain economic activities.

³¹ The F-test for the null of equality of coefficients is not rejected at the usual significance levels in this case.

of each quadrant in the panels is given by the proportion of observations falling in that quadrant, divided by the total number of observations. Therefore, a more intense background indicates that a greater proportion of observations fall this that quadrant. These panels seem to indicate that returns to RD&D investments hold a negative relationship with sectoral financial volatility (a proxy for uncertainty), and a positive, though weaker, correlation with the proportion of migrants holding university degree in the area (a proxy for regional spillovers).

In sum, as reported by the descriptive statistics for the national context presented in section 2, this section suggests that both returns to RD&D and market and knowledge factors differ significantly across sectors and regions. In the discussion that follows, we analyze the drivers for micro-level heterogeneity, assessing the role of context-based factors in boosting or lowering returns to innovative efforts.

5.4. The Role of Context-Based Complementary Factors

The challenge when analyzing the impact of potential complementary factors is the large list of candidates that can influence the returns to knowledge investments suggested by the literature summarized in section 3. To reduce this complexity, we built several indicators to proxy the six types of contextual factors described in section 3: STI policy; spillovers (sectoral and regional); demand; competition; regulation; and uncertainty. The details on how these indexes are constructed are summarized in table A.1 (appendix A).

To avoid including too many variables in the same regression—given that this can introduce collinearities and excessive loss of degrees of freedom—we first estimate six separate regressions, one for each of the groups of contextual factors (table A.2 of appendix A), as a way to explore them one at a time. From each regression, we chose the variable(s) that had a significant effect on productivity, with or without interacting with RD&D.³² We proceeded to include this subset of selected variables in single specifications, presented in table A.3. The table has eight columns because we had more than one candidate within some groups of contextual factors, which led us to a set of possible specifications. Among these options, our chosen estimation is the one presented in column 8. Appendix A explains why we consider this the best combination of contextual variables available.

Column 8 of table A.3 presents several interesting results. First, both **intra-sector spillovers** (expenditure on *innovation per sector/region*) and **demand** factors (*sector-region exports*) seem to be positively correlated with productivity levels (a double interaction), although not through RD&D investments (a triple interaction). The same holds for **competition**, measured by the Herfindahl index: a negative coefficient indicates that more concentrated markets are less productive on average, although the channel driving this relationship does not seem to be RD&D returns. Hence, these results suggest the existence of other unidentified channels that explain the relationship between these contextual factors and productivity. Examples of alternative channels could be that firms design other strategies apart from RD&D investments to absorb innovation spillovers or that higher levels of competition can lead managers to invest more in worker skills and human capital formation, which we are not measuring in our estimation.

 $^{^{32}}$ We did not produce IV estimates in this case because the IV procedure renders coefficients that are too large relative to FE in our knowledge production function estimation (table 1, column 7 versus column 4). In addition, it does not seem very feasible to find suitable instruments for all the necessary interactions. As a result, we focus primarily on describing the sign of the impact on the returns rather than the size effect, given the likely bias in the estimates.

In addition, **inter-sectoral spillovers** (*university migrants*) and our variable measuring market **uncertainty** are related to productivity through *RD&D returns*. For the former, a positive coefficient for the triple interaction indicates that investments in RD&D have a higher return on productivity among firms located in areas with a higher inflow of migrants with university degrees. This result is not surprising and could be led by the fact that a higher supply of educated workers increases RD&D returns, or more productive and innovative areas attract people with higher educational levels. Results for financial volatility are also as expected: sectors that suffer from more financial uncertainty have lower returns on innovation.

A puzzling result arises when **STI policy** (proxied by *Publications*) is interacted with RD&D. One potential explanation for this negative coefficient could be the existence of a substitution effect between firms doing in-house innovation versus those relying on external (public) knowledge embedded in publications relevant for their sector.

Finally, we find no significant relationship between productivity and **regulatory** measures. This can be the result of a lack of appropriate measurement, given that a vast literature suggests that the regulatory environment affects innovation incentives and results.

5.5. The Effect of Context-Based Factors via Innovative Capacity

The fact that in some cases contextual factors affect productivity directly, rather than through RD&D, suggests that innovative capacity or ability embedded in firm-level management processes and decisions may mediate the effect of contextual factors, mitigating or augmenting them.

We proxy this innovative capacity or ability using questions about managerial attitudes and skills in sections 2 and 7 from ENDEI 2. Specifically, we build indicators of managerial attitudes and skills that may account for firms' innovative capacity to benefit from *each* contextual factor. So, for example, an indicator that specifies which external sources of information the firm uses for innovation is relevant for the firm to be able to benefit, or not, from STI policies or spillovers. Similarly, firms that are technological leaders are more likely to be positively affected by contextual competition because greater competition will push them to increase their innovation activity to maintain their leadership.

To explore the role of these firm-level factors, the coefficients of interest are those that correspond to the *triple interaction* of RD&D, the context-based complementary factor, and the firm-level proxies of innovative capacity (in dealing with complementary factors). The level and significance of these estimators allows us to assess whether certain types of firms' attitudes make them better prepared to benefit from or to be held back by their context.

Table D.1 in appendix D describes each indicator built to account for firms' strategies (that is, innovative capacity) to benefit from each context-based complementary factor. In order to decide which of those managerial skills were actually relevant, we follow a procedure similar to the one used to select the contextual variables already described. First, we estimate separate regressions (table D.2). Then we select the significant ones for a second step where we include all of them together (table D.3). Results in table D.3 expand the set of conclusions derived from table A.3.

The results show that **STI policy** is now positively related to productivity for firms that rely on external sources of information. A feasible interpretation of this result is that firms are positively affected by being in an environment with higher supply of scientific articles if they have a strategy of pursuing learning from external sources. However, the coefficient multiplying the triple interaction of **STI policy**, RD&D, and our firm-specific dummy is negative and significant. This reinforces the potential existence of a substitution effect highlighted earlier. A greater external supply of information relevant for innovation may lead firms to rely less on internal RD&D procedures. Therefore, scientific knowledge has a positive effect on productivity for firms that use it (a positive double interaction), but is particularly important for firms that do not rely on internal efforts such as RD&D investments (a negative triple interaction).

Also, firms' attitudes toward external sources of information interact positively with intrasectoral **spillovers**, while the triple interaction remains insignificant. A likely interpretation for this result is that firms open to external sources of information benefit from intra-sectoral spillovers, regardless of whether they invest or not in RD&D.

Results regarding **demand** contextual factors show a positive relationship with productivity for all types of firms, although the channels driving this relationship differ. On the one hand, firms that declare demand to be a main driver for investment in innovation (variable DEM=1) seem to take better advantage of demand variables for their RD&D investments, consequently raising productivity (a positive triple interaction). On the other, for those firms that are not specifically motivated by demand for innovation, unspecified channels cause demand and firm productivity to positively correlate (a positive double interaction, DEM=0 * Exports).

Including interactions to account for firm' strategies also reveals new results for market **competition** factors. Firms that act in the market as technological leaders increase their returns to RD&D when the market is more competitive (the triple interaction is negative for Herfindahl*RD&D*Technoleader). This result is consistent with the predictions from Schumpterian theory (Aghion and Jaravel 2015). Competition also proves to be beneficial for the productivity of firms that are not technological leaders, as well, but through channels other than RD&D returns (a double interaction, Herfindahl*non-technological leader is negative). In fact, competition seems to harm the returns to RD&D for non-leaders.

While limited by our inability to measure some of these management strategies, the results suggest that innovative capacity, largely due to the quality of managerial practices, is likely to explain a portion of the observed heterogeneity in the returns to innovation, by means of enabling firms to take better advantage of context-based factors.

6. Conclusions and Implications for Policy

Main Findings

This paper presents new evidence to try to understand which factors are affecting the returns to innovation in Argentina, which can help address the dismally low volume of private R&D investment. The results show that returns are quite heterogeneous, which is likely to translate into the fairly low level of in-house private efforts to spur innovation and productivity growth. Our results suggest that different context-based factors, at the regional and sectoral levels, affect these returns. The findings also show that such effects are not homogeneous across firms

but depend on firms' strategies— referred to here as innovation capacity—which vary across firms.

Figure 10 summarizes which context-based complementary factors appear to have an impact on innovation returns (as shown in table A.3, column 8, and table D.3), . Results fall into three groups, according to the way they relate to firm productivity.

The first group consists of results associated with our first hypothesis (H1): those context-based complementary factors holding a relationship with RD&D returns on productivity for *all firms on average*. A second set of results are those associated with our second hypothesis (H2): those context-based complementary factors holding a relationship with productivity through the RD&D channel, but *only for firms with specific innovative capacity* (that is, proactive attitudes to make the most of context-based complementary factors). In addition, a third group of results, not addressed by these two hypotheses, includes some context-based factors relating to productivity, although not through RD&D investments.

The first group refers to H1: context-based factors that are significantly correlated with the returns to RD&D whatever the strategic action/attitude toward the context the firm has. We find that H1 is not rejected for **uncertainty**, which affects returns negatively, and **intersectoral spillovers**, which correlates positively. Results for both factors are as expected: more volatile contexts do not favor returns to innovation investments, while inter-sectoral spillovers enhance returns to RD&D. These are robust findings given that both variables are also statistically significant (p-value < 1% for uncertainty, p-value < 5% for spillovers) in all specifications. Hence, these two contextual factors should be particularly addressed by policies looking to boost innovation returns because, on average, they seem to have an effect on all firms.

The second group relates to **H2:** context-based complementary factors that have a significant relationship with productivity through RD&D returns, but only when mediated by some proactive attitudes by firms. We find some empirically validity for competition and demand. In the case of market **competition**, we find that it affects innovation returns positively only when mediated by the strategic position of the firm in the market (whether it is a technological leader or not). This result is important for policy because it seems to indicate that high levels of competition favor innovation of technological leaders, as has been found by the Schumpeterian literature. Policies need to consider this heterogeneous impact of competition because competition lowers the returns to RD&D investment for firms that are not leaders in their markets. A similar result appears for the **demand** factor, which is significantly related to RD&D returns only when mediated by firms' attitudes toward demand: greater demand orientation in their strategies can lead to greater investment in innovation. An unexpected result is the negative effect of **STI policy**, proxied by industry-relevant publications, on RD&D returns for firms that are particularly active in searching academic sources of external knowledge.

The third group includes context-based factors that affect productivity, but not through RD&D returns. **Intra-sectoral spillovers** belong to this group. When mediated by how open the firm is toward the context (that is, how many market-related external sources of information the firm uses), this factor affects productivity through other unspecified channels.

In addition, the **STI policy** factor yields interesting conclusions. Firms that consult sources from academic organizations increase their productivity when publications relevant for their

industrial activity are abundant (the STI system is favorable in this regard). However, as noted, the effect through the RD&D channel is negative. This suggests a substitution effect, by which firms either rely on in-house RD&D or are nurtured by the STI environment.

Finally, we do not have enough evidence to establish a relevant relationship between **regulation** factors and productivity, either through RD&D or through any other channel. This could be due to problems in measuring these regulatory factors.

A caveat is in order in drawing conclusions from these results. Robust assessment of the causal relationship between context-based complementary factors, RD&D, and productivity is difficult to achieve with the data available. The results shown are fairly stable and we are confident that panel data estimations have minimized endogeneity issues greatly. However, instrumental variables or other quasi-experimental methods would be necessary to ensure causal interpretations of the coefficients. This could be an interesting avenue for future research with better data sets. Appendix F presents the implications from this research related to data collection and offers suggestions for improvements.



Figure 10. Synthesis of Results

Implications for Policy

The findings on the relationship between context-based and complementary factors, firmspecific strategies, and returns to investments in innovation have important implications for public policy. First, there is a need for policy to address some of the contextual factors discussed. For instance, political programs looking to provide incentives for innovation should focus on supporting firms' R&D investment in periods of economic uncertainty or promoting mobility of academics and experts across the country to favor intersectoral spillovers. This contrasts with the current view of encouraging R&D efforts via the use of tax incentives. Such incentives are procyclical by design, given that they depend on the provisions to accumulate tax credits, which can amplify the effect of uncertainty. The heterogeneous impact of competition is also relevant. Policy measures that encourage innovation in laggards could be useful in the presence of significant market failures.

The second important implication for policy is related to the relevance of regional innovation strategies. The impact of these contextual factors as complementarities to R&D and the fact that many of these factors are regional in nature suggest the importance of having flexible STI policies that adapt to the needs of the local innovation system. This recommendation is supported by the concept of "smart specialization" (Foray, David, and Hall 2011). This strategy has gained increasing popularity in Europe recently. It consists of a territorial-based setting of priorities that involves different stakeholders from academia, business organizations, policy authorities, and civil society. This is done through a process called "entrepreneurial discovery," which consists of identifying investment priorities largely grounded on local assets and resources ((Gomez Prieto, Demblans, and Palazuelos Martinez 2019). Strategies that are developed from the center with little adaptation will likely fail in addressing critical failures and the impact of contextual factors. This more decentralized approach will require "federalization" of public innovation programs and adaptability. This could be achieved by transferring the design and implementation of certain innovation policies to provincial ministries-those with greater government capabilities-and improving communication between local firms; research organizations that have regional headquarters such as CONICET (the National Scientific and Technical Research Council), INTA (the National Agricultural Technology Institute), and INTI (the National Industrial Technology Institute); regional universities; and civil society organizations.

The third critical policy implication is the need to invest in innovation capacity. Having better quality management strategies is more likely to maximize the impact of these context-based factors on the returns to R&D. For example, building these management capabilities to maximize external knowledge, or to appropriate the rewards of innovation in specific contexts, or to use and recruit talent, can be critical to maximize the impact of innovation policies. Instruments that support this building of managerial capabilities have a central role to play in innovation policy to build more sophisticated innovation policy support and should be mainstreamed in STI policies.

7. References

Acemoglu, Daron. 2003. "Root causes." Finance & Development no. 40 (2):27-43.

- Acemoglu, Daron, Ufuk Akcigit, Harun Alp, Nicholas Bloom, and William Kerr. 2018. "Innovation, reallocation, and growth." *American Economic Review* no. 108 (11):3450-91.
- Acemoglu, Daron, Simon Johnson, James Robinson, and Yunyong Thaicharoen. 2003. "Institutional causes, macroeconomic symptoms: volatility, crises and growth." *Journal of monetary economics* no. 50 (1):49-123.
- Adner, Ron. 2002. "When are technologies disruptive? A demand-based view of the emergence of competition." *Strategic Management Journal* no. 23 (8):667-688.
- Aghion, P., N. Bloom, R. Blundell, R. Griffith, and P. Howitt. 2005. "Competition and Innovation: An Inverted-U Relationship." *Quarterly Journal of Economics* no. 120 (2):701-728.
- Aghion, P., and X. Jaravel. 2015. "Knowledge spillovers, innovation and growth." *Economic Journal* no. 125 (583):533-573. doi: 10.1111/ecoj.12199.
- Antoncic, Bostjan, Igor Prodan, Robert D. Hisrich, and Cezar Scarlat. 2007. "Technological Innovativeness and Firm Performance in Slovenia and Romania." *Post-Communist Economies* no. 19 (3):281-298.
- Arza, Valeria. 2013. "The Social Dimension of Behaviour: Macroeconomic Uncertainty and Firms' Investment in R&D and in Machinery in Argentina." In *Learning, Capability Building and Innovation for Development*, 176-201. Palgrave Macmillan, London.
- Arza, Valeria, and Andrés López. 2010. Innovation and productivity in the argentine manufacturing sector. In *IDB Working Paper No. 62*.
- Arza, Valeria, and Claudia Vázquez. 2014. Evaluación del diferencial en el aumento de inversión en actividades innovativas respecto a ventas entre empresas beneficiarias del Programa de Innovación Tecnológica vs grupo control, 2008–2012. In Consultora BSI-Aguilar. Unpublished.
- Arza, Valeria, and Claudia Vázquez. 2015. Evaluación del diferencial en el aumento de inversión en actividades innovativas respecto a ventas entre empresas beneficiarias del Programa de Innovación Tecnológica vs grupo control, 2008–2013. In Consultora BSI-Aguilar. Unpublished.
- Audretsch, Bruce. 1998. "Agglomeration and the location of innovative activity." Oxford review of economic policy no. 14 (2):18-29.
- Benavente, Jose Miguel. 2006. "The role of research and innovation in promoting productivity in chile." *Economics of Innovation & New Technology* no. 15 (4/5):301-315.
- Bernanke, Ben S. 1983. "Irreversibility, uncertainty, and cyclical investment." *The Quarterly Journal* of *Economics* no. 98 (1):85-106.
- Bloom, Nicholas. 2009. "The impact of uncertainty shocks." Econometrica no. 77 (3):623-685.
- Bloom, Nicholas, Charles I Jones, John Van Reenen, and Michael Webb. 2017. Are ideas getting harder to find? : National Bureau of Economic Research.
- Brambilla, Irene, and Darío Tortarolo. 2018. Investment in ICT, productivity, and labor demand: the case of Argentina: The World Bank.
- Caballero, R. J., and R. S. Pindyck. 1996. "Uncertainty, Investment. and Industry Evolution." *International Economic Review* no. 37 (3):641-662.
- Cirera, X., W.F. Maloney, and M. Sarrias. 2019. "Management Quality as an Input for Innovation." *World Bank mimeo*.
- Cooke, Philip. 2001. "Regional innovation systems, clusters, and the knowledge economy." *Industrial and corporate change* no. 10 (4):945-974.
- Cooke, Philip, Mikel Gomez Uranga, and Goio Etxebarria. 1997. "Regional innovation systems: Institutional and organisational dimensions." *Research policy* no. 26 (4-5):475-491.
- Cox, Peter J. 1990. "Research and development or research design and development?" *International Journal of Project Management* no. 8 (3):144-150. doi: https://doi.org/10.1016/0263-7863(90)90015-4.

- Crepon, Bruno, Emmanuel Duguet, and Jacques Mairesse. 1998. "Research, Innovation And Productivity: An Econometric Analysis At The Firm Level." *Economics of Innovation and new Technology* no. 7 (1):115-158.
- Crescenzi, R., and A. Rodríguez-Pose. 2013. "R&D, socio-economic conditions, and regional innovation in the U.S." *Growth and Change* no. 44 (2):287-320. doi: 10.1111/grow.12011.
- Crespi, Gustavo, Ezequiel Tacsir, and Fernando Vargas. 2016. "Innovation dynamics and productivity: Evidence for Latin America." In *Firm Innovation and Productivity in Latin America and the Caribbean*, 37-71. Springer.
- Crespi, Gustavo, and Pluvia Zuniga. 2012. "Innovation and productivity: evidence from six Latin American countries." *World development* no. 40 (2):273-290.
- Chudnovsky, Daniel. 2001. Innovation activities by manufacturing firms in a more open developing economy: the case of Argentina in the 1990s. Paper read at Aalsborg, Denmark: Nelson and Winter Conference, No. 45.
- Chudnovsky, Daniel, Andrés López, and Germán Pupato. 2004. "Innovation and productivity in developing countries: A study of Argentine manufacturing firms' behaviour (1992-2001)." *Mimeo, presented to Research Policy*.
- Davies, Stephen, and Bruce Lyons. 1996. "Chapter 6 Industrial Concentration." In Industrial Organization in the European Union. Oxford: Clarendon Press.
- de Elejalde, Sheila, David Giuliodori, and Rodolfo Stucchi. 2011. Employment Generation, Firm Size and Innovation: Microeconometric evidence from Argentina. Inter-American Development Bank.
- Dixon, R., and A. P. Thirlwall. 1975. "A Model of Regional Growth-Rate Differences on Kaldorian Lines." *Oxford Economic Papers* no. 27 (2):201-14.
- Doraszelski, Ulrich, and Jordi Jaumandreu. 2013. "R&D and productivity: Estimating endogenous productivity." *Review of Economic Studies* no. 80 (4):1338-1383.
- Dosi, Giovanni. 1988. "Sources, procedures, and microeconomic effects of innovation." Journal of economic literature:1120-1171.
- Fanelli, Jose Maria. 2002. "Growth, instability and the convertibility crisis in Argentina." *Cepal Review* (77):25–43.
- Fares, Florencia M., Guido Zack, and Ricardo Martínez. 2017. "Cálculo de los índices de precio sectoriales anuales de las importaciones argentinas. Metodologías y usos." *REVISTA ECONOMÍA Y DESAFÍOS DEL DESARROLLO* no. Volúmen I (Número 3).
- Foray, Dominique, Paul A. David, and Bronwyn H. Hall. 2011. Smart specialisation From academic idea to political instrument, the surprising career of a concept and the difficulties involved in its implementation. In *MTEI Working Paper, École Polytechnique Fédérale de Lausanne*. MTEI-WORKING_PAPER-2011-001.
- Frank, Andrew Gunder. 1959. "Industrial capital stocks and energy consumption." *The Economic Journal* no. 69 (273):170-174.
- Fung, M. K. 2004. "Technological opportunity and productivity of R&D activities." Journal of Productivity Analysis no. 21 (2):167-181. doi: 10.1023/B:PROD.0000016871.95892.41.
- Galiani, Sebastian, Daniel Gomez, and Gabriel Scattolo. 2019. Productivity in Argentina. In Background paper to the World Bank Argentina Productivity Project.
- García-Quevedo, José, Gabriele Pellegrino, and Maria Savona. 2016. "Reviving demand-pull perspectives: The effect of demand uncertainty and stagnancy on R&D strategy." *Cambridge Journal of Economics* no. 41 (4):1087-1122.
- Gomez Prieto, Javier, Albane Demblans, and Manuel Palazuelos Martinez. 2019. Smart Specialisation in the world, an EU policy approach helping to discover innovation globally: Outcomes, lessons and reflections from the first global workshop on Smart Specialisation. No. JRC117005. Joint Research Centre (Seville site).
- Goñi, E., and W. F. Maloney. 2017. "Why don't poor countries do R&D? Varying rates of factor returns across the development process." *European Economic Review* no. 94:126-147. doi: 10.1016/j.euroecorev.2017.01.008.
- Griffith, Rachel, Stephen Redding, and John Van Reenen. 2004. "Mapping the two faces of R&D: Productivity growth in a panel of OECD industries." *Review of economics and statistics* no. 86 (4):883-895.

Griliches, Zvi. 1967. "Distributed lags: A survey." Econometrica (pre-1986) no. 35 (1):16.

- Griliches, Zvi. 1979. "Issues in Assessing the Contribution of Research and Development to Productivity Growth." *The Bell Jorunal of Economics* no. 10 (1):92-116.
- Griliches, Zvi. 1994. "Productivity, R&D, and the Data Constraint." *American Economic Review* no. 84 (1):1-23.
- Harrigan, James, Ariell Reshef, and Farid Toubal. 2018. "Techies, Trade, and Skill-Biased Productivity." *National Bureau of Economic Research Working Paper Series* no. No. 25295. doi: 10.3386/w25295.
- Jefferson, Gary H., Huamao Bai, Xiaojing Guan, and Xiaoyun Yu. 2006. "R&D Performance in Chinese industry." *Economics of Innovation & New Technology* no. 15 (4/5):345-366.
- Kafouros, M. I., and P. J. Buckley. 2008. "Under what conditions do firms benefit from the research efforts of other organizations?" *Research Policy* no. 37 (2):225-239. doi: 10.1016/j.respol.2007.11.005.
- Kaldor, Nicholas. 1966. Causes of the slow rate of economic growth of the United Kingdom: an inaugural lecture: Cambridge University Press.
- Katz, Jorge. 1982. Cambio tecnológico en la industria metalmecánica latinoamericana: resultados de un programa de estudios de casos. Inter-American Development Bank.
- Katz, Jorge. 2000. "Pasado y presente del comportamiento tecnológico de América Latina." *Revista de la CEPAL* no. Serie de Desarrollo Productivo (75).
- Katz, Jorge, and Gonzalo Bernat. 2011. "Exit-entry, productivity growth and structural change in response to changes in macroeconomic policy: evidence from Argentina." *Innovation and Development* no. 1 (2):227-244.
- Klenow, Peter J, and Andres Rodriguez-Clare. 2005. "Externalities and growth." *Handbook of economic* growth no. 1:817-861.
- Klevorick, Alvin K, Richard C Levin, Richard R Nelson, and Sidney G Winter. 1995. "On the sources and significance of interindustry differences in technological opportunities." *Research policy* no. 24 (2):185-205.
- Kline, SJ, and N Rosenberg. 1986. "N.(1986): An overview of innovation." *The Positive Sum Game, Washington. DC*.
- Kosacoff, Bernardo. 2000. Corporate strategies under structural adjustment in Argentina: Responses by industrial firms to a new set of uncertainties: New York: St Martin's Press.
- Krugman, Paul R. 1991. Geography and trade: MIT press.
- Lundvall, Bengt-Ake. 1992. National Innovation Systems: Towards a Theory of Innovation and Interactive Learning. London: Pinter.
- Malerba, F. 2002. "Sectoral systems of innovation and production." *Research Policy* no. 31 (2):247-264. doi: 10.1016/S0048-7333(01)00139-1.
- Malerba, F., and L. Orsenigo. 1995. "Schumpeterian Patterns of Innovation." *Cambridge Journal of Economics* no. 19 (1):47-65.
- Maloney, W. F. 2017. "Revisiting the national innovation system in developing countries."
- Maloney, William, and Andrés Rodríguez-Clare. 2007. "Innovation shortfalls." *Review of Development Economics* no. 11 (4):665-684.
- Mansfield, E. 1972. "Contribution of R&D to economic growth in the United States." *Science* no. 175 (4021):477-486. doi: 10.1126/science.175.4021.477.
- Marín, A., and S. Petralia. 2018. "Sources and contexts of inter-industry differences in technological opportunities: The cases of Argentina and Brazil." *Innovation and Development* no. 8 (1):29-57. doi: 10.1080/2157930X.2018.1429191.
- Marshall, Alfred. 1890. *Principles of Economics*. Reprint 1961 ed: London; New York: Macmillan for the Royal Economic Society.
- McCombie, John, and Tony Thirlwall. 1995. Economic Growth and the Balance-of-Payments Constraint. New York: St Martin's Press.
- Metcalfe, Stan, and Ronnie Ramlogan. 2008. "Innovation systems and the competitive process in developing economies." *The Quarterly Review of Economics and Finance* no. 48 (2):433-446.
- Nelson, Richard R. 1993. *National innovation systems: a comparative analysis*: Oxford University Press on Demand.

- Ngai, L. R., and R. M. Samaniego. 2011. "Accounting for research and productivity growth across industries." *Review of Economic Dynamics* no. 14 (3):475-495. doi: 10.1016/j.red.2009.12.002.
- OECD. 1998. "STI Review: Special Issue on "New Rationale and Approaches in Technology and Innovation Policy" No. 22 Volume 1998 Issue 1." *STI Science, Technology Industry Review* no. 1998 (1):1-315.
- Olea, José Luis Montiel, and Carolin Pflueger. 2013. "A robust test for weak instruments." *Journal of Business & Economic Statistics* no. 31 (3):358-369.
- Parisi, Maria Laura, Fabio Schiantarelli, and Alessandro Sembenelli. 2006. "Productivity, innovation and R&D: Micro evidence for Italy." *European Economic Review* no. 50 (8):2037-2061.
- Pavitt, Keith. 1984. "Sectoral patterns of technical change: Towards a taxonomy and a theory." *Research Policy* no. 13 (6):343-373.
- Pereira, Mariano, and Diana Suárez. 2018. "Matthew effect, capabilities and innovation policy: the Argentinean case." *Economics of Innovation and New Technology* no. 27 (1):62-79.
- Pindyck, R. S., and A. Solimano. 1993. "Economic Instability and Aggregate Investment." Nber Working Paper Series (4380):ALL.
- Roper, Stephen, and James H Love. 2018. "Knowledge context, learning and innovation: an integrating framework." *Industry and Innovation* no. 25 (4):339-364.
- Sargan, J. D. 1958. "The Estimation of Economic Relationships using Instrumental Variables." *Econometrica* no. 26 (3):393-415. doi: 10.2307/1907619.
- Scherer, Frederic M. 1967. "Research and development resource allocation under rivalry." *The Quarterly Journal of Economics* no. 81 (3):359-394.
- Schmookler, Jacob. 1962. "Economic sources of inventive activity." *The Journal of Economic History* no. 22 (1):1-20.
- Schumpeter, Joseph A. 1942. Capitalism, socialism, and democracy. Reprint 2008 ed: Harper Collins.
- Siliverstovs, Boriss. 2016. "R&D and non-linear productivity growth." *Research Policy* no. 45 (3):634-646.
- Soete, Luc, Bart Verspagen, and Bas Ter Weel. 2010. "Systems of innovation." In *Handbook of the Economics of Innovation*, 1159-1180. Elsevier.
- Strobel, Thomas. 2012. "New evidence on the sources of EU countries' productivity growth—industry growth differences from R&D and competition." *Empirica* no. 39 (3):293-325.
- Suárez, Diana, Gustavo Lugones, and Pablo Moldovan. 2008. Innovation, competitiveness and salaries: A model of combined growth at the firm level. Paper read at Globelics Mexico 2008, November 22-24, at Mexico D. F.
- Sutton, John. 1998. *Technology and market structure : theory and history*. Cambridge, Mass: The MIT Press.
- Teixeira, Aurora AC, and Anabela SS Queirós. 2016. "Economic growth, human capital and structural change: A dynamic panel data analysis." *Research policy* no. 45 (8):1636-1648.

Appendix A. Choosing among Contextual Variables

Table A.1. Explanation of Contextual Variables

Contextual factor	Variable	Variable name in tables	Explanation	Intends to measure	Period of data availability	Data source
	Academic publications per sector	Publications	Built from Scopus microdata on academic publications. Scopus assigns each publication a field of study. To convert this to sectoral-level data, we use a matrix built from ENIT(National Survey on Innovation and Technology Behavior) data, which offers information on how much each sector of economic activity values each field of study.	Supply of scientific knowledge per sector	2013–17	Scopus + ENIT
Saiamaa	Proportion of people with university degree per province (EPH)	PropUniv	Proportion of inhabitants holding a university degree per province.	Supply of qualified workforce per province	2003–18	ЕРН
technology, and innovation (STI) policy	University graduates per sector	GraduatesSec	Built from micro-level data on university graduates. To convert career data to sectoral-level information, we first group careers into fields of study. We then use the same matrix as with the Scopus data to convert this to sectoral-level data.	Supply of qualified workforce per sector	2010–17	Department of universitary information
	Provincial budget for higher education	HigherEduc	Public expenditure on university- and tertiary-level education as a proportion of total public expenditure by province.	Government support for building qualified workforce	2001–16	Ministry of Education
	PCA (principal component analysis) of provincial budgets	PCAbudget	1st component resulting from principal component analysis (PCA). Variables included are budget amounts designated for different areas considered relevant for innovation: education and culture; primary, secondary and tertiary education; and private spending on education.	Government support for R&D activities	2001–16	National Directorate of Provincial Affairs, Sub-Secretariat of Provincial Coordination
	Innovation per sector- region	Inno_sector_region	Results from adding up investment in any innovation activity for firms within same sector and region.	Intrasectoral spillovers	2014–16	ENDEI
Spillovers– sectoral	Intersectoral spillovers	Inno_spillover	Weighted sum of the investment in innovation in other sectors. The weights for this sum represent the importance of various sectors for a specific sector. To calculate these weights, we construct a matrix with the movement of employees between sectors across time in Argentina using EPH data. Therefore, sector A will be more relevant for another sector B if the latter has many employees who worked in sector	Spillovers of investment in innovation coming from other sectors.	2014–16	ENDEI+EPH

			A in the past (and are bringing their knowledge and experience to sector B).			
Spillovers- regional (spatial)	Proportion of inmigrants with university degree	MigUniv	Proportion of inhabitants of a province that are immigrants (originally from another country or province) and hold a university degree (interprovince spillovers).	Movement of qualified workforce (inter-sectoral spillovers)	2003–18	ЕРН
	Exports sector-province	Exports	Value of yearly exports in dollars by province and sector. Prices adjusted with sectoral level price index.	Level of international demand	1997–2017	OPEX; Fares, Zack, and Martínez (2017) for price adjustments
Demand	Unemployment rate	UnempRate	Unemployed population/Economically active population per province.	Level of demand in province	2003-18	EPH
	Mean wages	MeanSal	Mean income from main occupation per province.	Level of demand in province	2003–18	EPH
	Mean household income per capita	HHinc	Mean household income per capita by province.	Level of demand in province	2003–18	EPH
	Herfindahl index (sector-region)	Herfindahl	Sector-Region Herfindahl index constructed with firm level sales data from ENDEI.	Level of competition in sector-region	2014–16	ENDEI
~	Number of firms (province-sector)	NumFirms	Number of operating firms in province-sector.	Level of competition in sector-province (when this proxy is used, we control by sector-province size using total employment)	1996–16	OEDE
Competition	Average firm size (province-sector)	FirmSize	Total employment/Number of firms at province- sector level.	Level of competition in sector-province	1996–16	OEDE
	Employment sector- province	Employment sector-province	Total number of employees per sector-province.	Control for market size when including number of firms as competition variable	1996–2018	OEDE
	Proportion of people in informal sector	PropInformal_SEC	Proportion of employees in each sector stating they are not contributing to a formal pension system.	Intends to measure regulation in sector. Higher informality is a sign of less regulation	2003–18	EPH
Regulation	Google searches on corruption topics (province)	Corruption_GT	Google trends index on searches of words related to corruption by province. This index is constructed in a way to control for the total amount of google searches per province and period of time.	Level of corruption per province. Assumes that inhabitants of provinces with more corruption tend to search more on corruption related topics	2004–19	Google Trends

	Opening + Closing of firms	FirmEntryExit	(# Firm Entry+ # Firm exit)/ # of Firms per province.	Represents dynamism of industry in each province. A higher value of this variable means that there are less regulations for opening and closing a business	1996–2016	OEDE
	Proportion of people affiliated to a political party	PropAffil	Proportion of voting population affiliated to a political party per province	Measure of accountability for each province. Higher levels of affiliation represent a greater control of the state by the population, which tends to lead to better economic institutions.	2013–17	National electoral chamber
	Wage deviation by sector (EPH)	WageDev	Coefficient of variation of individual level data on wages per sector	Attempts to measure power of trade unions in each sector, as these tend to decrease the dispersion of salaries.	2003–18	ЕРН
	Volatility of financial credits (per sector)	Volat_credit_sect	Std deviation of residuals of AR(1) model using data on financial credits at sectoral level		2000–17	BCRA
	Volatility of financial credits (per province)	Volat_credit_prov	Std deviation of residuals of AR(1) model using data on financial credits at province level		2000-17	BCRA
	Volatility of imports	Volat_imports	Std deviation of residuals of AR(1) model using data on sectoral imports		1962–2018	COMTRADE
	Volatility of interest rate	Volat_intrate	Std deviation of residuals of AR(1) model using data on sectoral level nominal interest rates	Market uncertainty. A higher variance in the residuals	2000–17	BCRA
Uncertainty	Volatility of export price	Volat_priceX	Std deviation of residuals of AR(1) model of sectoral level export prices	of an autoregressive process indicates that the variable is less predictable.	1996–2016	Fares, Zack, and Martínez (2017)
	Volatility of exports (per province)	Volat_expo_prov	Std deviation of residuals of AR(1) model using data on exports at province level		1997–2017	OPEX
	Volatility of exports (per sector)	Volat_expo_sect	Std deviation of residuals of AR(1) model using data on exports at sectoral level		1997–2017	OPEX
	Volatility of exchange rate (per sector)	Volat_ER	Std deviation of residuals of AR(1) model using data on exchange rate at sectoral level		2010–18	Secretariat of Productive Transformation

Note: BCRA = Central Bank of Argentina ; COMTRADE = United Nations International Statistics database; ENDEI = Employment and Innovation Dynamics National Survey; ENIT = National Survey on Innovation and Technology Behavior, EPH = Permanent Household Survey ; OEDE = Observatory of Labor and Firm Dynamics; OPEX = Provincial origin of Argentine Exports

In table A.2, each column presents estimations for the production function, including all indicators built to account for each context-based factor, using the specification in column 2 from table 1 in the main text. Indicators that are statistically significant are boldfaced.

Based on significant variables from table A.2, we then estimated many alternative regressions taking into account variables that were highly correlated, and situations when more than one variable was significant within the same group (recall that our interest lies in getting a good proxy variable for the contextual factor that it represents). Results can be seen in table A.3.

Column 1 includes one significant variable from each context-based factor, excluding regulation because there was not any relevant result in that case. Column 2 includes the proxy for regulation that we are more confident about (entry plus exit of firms, which suggests bureaucratic ease to promote business dynamism).

Since coefficients do not change much, for the sake of completeness we decided in favor of column 2 (including regulation). In column 3, we changed our proxy for STI policy, using university graduates per sector instead of academic publication per sector. Akaike information criterion (AIC), Bayesian information criterion (BIC), and the significance of coefficients drive us to choose column 2.

Column 4 changes the proxy for spillovers; instead of using the proportion of university immigrants in the region, which accounts for inter-sectoral spillovers, we include the average innovation expenditures per sector region, which accounts for intra-sector spillovers. Although the variable for intra-sectoral spillover is significant, results are marginally better for the former, in terms of AIC and BIC criterion, so we stick to results in column 2.

In column 5 we change the proxy for competition; instead of using the number of firms in the sector-province, we use the Herfindahl index. Results improve, according to AIC and BIC criterion, leading us to choose in favor of the specification in column 5.

Columns 6 and 7 change our proxy for uncertainty, using volatility of exports and imports, respectively, instead of volatility of credit resources. Results do not improve, so we stick to column 5.

Finally, in column 8, we include both variables for spillovers (inter-sectoral and intra-sectoral) that we have tested for separately, and results improve. Since they conceptually account for different aspects of spillovers, both remain significant. AIC criteria suggest this is the best specification, so we opt for that one to continue the analysis.

Table A.2. Results for Equation [2], Including Several Proxies of Complementary Factors

Each column presents one specification for each of the six complementary factors identified in figure 5

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	STI policy	Spillovers	Demand	Competition	Regulation	Uncertainty
Capital per worker	0 30***	0 30***	0 30***	0 30***	0 30***	0 30***
	(0.032)	(0.032)	(0.032)	(0.032)	(0.032)	(0.032)
R&D+D per worker	-0.0036	-0.0010	0.0065	0.0042	0.00010	0.0077
	(0.020)	(0.0054)	(0.011)	(0.0049)	(0.012)	(0.0047)
Labor	-0.26^{***}	-0.26^{***}	-0.26^{***}	-0.25^{***}	-0.26***	-0.26^{***}
Year 2015 dummy	0.20***	0.18***	0.18***	0.18***	0.18***	0.17***
	(0.018)	(0.017)	(0.019)	(0.015)	(0.017)	(0.018)
Year 2016 dummy	0.35***	0.34***	0.32***	0.35***	0.34***	0.33***
	(0.031)	(0.031)	(0.050)	(0.023)	(0.024)	(0.028)
Missing dummy	0.10^{**}	0.085^{*}	0.091^{*}	0.099^{**} (0.048)	0.095** (0.047)	0.098^{**} (0.047)
Academic publications per sector	19.6**	(0.010)	(0.010)	(0.010)	(0.017)	(0.017)
······································	(8.16)					
Prop of people with university degree per prov	-1.37					
	(1.41)					
University graduates per sector	-1.38					
Provincial Budget for higher education	(4.49)					
Trovincial Budget for inglier education	(0.34)					
PCA of provincial budgets	-0.025					
	(0.022)					
RD&D p.w. * Publications	-0.14					
	(0.24)					
RD&D p.w. * PropUniv	(0.21)					
RD&D p.w. * GraduatesSec	-0.23					
1	(0.21)					
RD&D p.w. * HigherEduc	-0.0018					
	(0.049)					
RD&D p.w. * PCA Budget	(0.00085)					
Innovation per sector-region	(0.0028)	3.04**				
1 8		(1.18)				
Intersectoral spillovers		-0.37				
		(1.08)				
Prop of initigrants with univer degree		(2.89)				
RD&D p.w. * Inno sector region		-0.14				
1 0		(0.11)				
RD&D p.w. * Inno_spillover		0.088				
		(0.087)				
RD&D p.w. * MigUniv		0.36**				
Exports sector-province		(0.13)	0.016***			
			(0.0055)			
Unemployment rate			1.10			
			(1.09)			
Mean salary			1.33			
Mean household income per capita			(7.15)			
Wear nousehold meome per capita			(6.18)			
RD&D p.w. * Exports			0.00017			
			(0.00046)			
RD&D p.w. * UnempRate			0.041			
			(0.14)			
KD&D p.w. * MeanSalar			0.39			
RD&D p.w. * HHincome			-0.43			
			(0.39)			
Herfindahl index			· · · ·	-0.00010**		
				(0.000040)		
Number of firms				-7.82		
Average firm size				-4.0e-09		
2						

RD&D p.w. * Herfindahl				(1.1e-08) 1.7e-06		
RD&D p.w. * NumFirm				(2.0e-06) 0.41 *		
RD&D p.w. * Firm size				(0.22) -1.9e-10		
Employment sector-province				(5.7e-10) -4.8e-06 (9.8e-06)		
Proportion of people in informal sector				(9.86-86)	0.061	
Google searched on corruption topics					0.00083	
Entry+ Exit of firms					(0.00055) 0.37	
Prop of people affiliated to a political party					(0.62) 0.056	
Wage deviation by sector (EPH)					(0.039) -0.042*	
RD&D p.w. * PropInformal_SEC					(0.025) 0.0060	
RD&D p.w. * Corruption_GT					(0.0097) 8.2e-06	
RD&D p.w. * FirmEntryExit					(0.000054) 0.090	
RD&D p.w. * PropAffil					(0.072) -0.0040	
RD&D p.w. * WageDev					(0.0040) -0.0060 (0.0040)	
Volatility of financial credits (sector)					/	97.7
Volatility of imports (sector)						-616 (423)
Volatility of interest rate (sector)						58.1
Volatility of export price (sector)						(/3.8) -190***
Volatility of financial credits (province)						(54.5) -47.0
Volatility of exports (province)						(04.8) -103
Volatility of exports (sector)						(67.7) 123
Volatility of exchange rate (sector)						-0.0062
RD&D p.w. * Volatility of imports (sector)						(0.0069) -17.6
RD&D p.w. * Volatility of interest rate (sector)						(13.7) 10.7
RD&D p.w. * Volatility of financial credits (province)						(7.73) -35.1***
RD&D p.w. * Volatility of export price (sector)						(12.2) 1.37 (11.7)
RD&D p.w. * Volatility of financial credits (province)						(11.7) 3.52
RD&D p.w. * Volatility of exports (province)						(10.5) 5.68
RD&D p.w. * Volatility of exports (sector)						(9.50) -2.05
RD&D p.w. * Volatility of exchange rate (sector)						(11.8) 0.00061 (0.00084)
Constant	10.5***	10.6***	10.5***	10.9***	10.6***	10.7***
Observations R-squared Number of ID	9,246 0.391 3,489	9,246 0.391 3,489	9,246 0.391 3,489	9,176 0.390 3,471	9,246 0.391 3,489	9,246 0.393 3,489

Note: Indicators that are statistically significant are boldfaced. EPH = Permanent Household Survey; PCA = principal component analysis; p.w. = per worker; R&D +D = research design and development; STI = science, technology, and innovation. Clustered standard errors in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

Contextual Factor	VARIABLES	(1) No regulatio n	(2) Including Regulation	(3) Change STI policy	(4) Change Spillovers	(5) Change competiti on	(6) Change Uncert	(7) Change Uncert	(8) Both Spillovers
	Capital per worker	0.30***	0.30***	0.30***	0.30***	0.30***	0.30***	0.30***	0.30***
	R&D+D per worker	(0.032) 0.0022	(0.032) -0.0075	(0.032) -0.0039	(0.032) 0.0022	(0.032) -0.0040	(0.032) -0.0044	(0.032) -0.0043	(0.032) -0.0030
	Labor	(0.0060) -0.26***	(0.012) -0.26***	(0.013) -0.26***	(0.011) -0.26***	(0.012) -0.26***	(0.012) -0.26***	(0.012) -0.26***	(0.012) -0.26***
	Year =2015	(0.045) 0.20***	(0.045) 0.20***	(0.045) 0.19^{***}	(0.045) 0.19***	(0.046) 0.19***	(0.046) 0.18^{***}	(0.045) 0.19***	(0.046) 0.19^{***}
	Year =2016	(0.016) 0.32***	(0.016) 0.31***	(0.016) 0.35***	(0.016) 0.32***	(0.015) 0.33***	(0.015) 0.33^{***}	(0.015) 0.33***	(0.015) 0.32^{***}
	Missing dummy	(0.027) 0.099**	(0.028) 0.100** (0.048)	(0.026) 0.100** (0.048)	(0.027) 0.10**	(0.028) 0.098** (0.047)	(0.029) 0.098** (0.048)	(0.028) 0.096^{**}	(0.029) 0.100**
STI policy	Acad. publications per sector	21.1**	21.2***	(0.048)	<u>(0.050)</u> 15.0*	<u>(0.047)</u> 15.9**	13.4*	13.2	<u>(0.047)</u> 8.81
1 2	······ F ······· F ·· ·····	(8.07)	(8.03)	-	(8.99)	(7.49)	(7.86)	(8.02)	(8.50)
STI policy	University graduates per sector	-	-	-1.69		-	-	-	-
Spillovers	Prop of inmig with univer degree	- 1 91	- 2 64	(4.11) 2.76	_	- 2 64	- 2 65	- 2 62	- 2 78
Spinovers	The point ming with univer degree	(2.90)	(3.24)	(3.25)	-	(3.19)	(3.18)	(3.20)	(3.19)
Spillovers	Innovation per sector-region	-	-	-	2.83**	-	-	-	3.08***
Demand	Exports	-	- 0.017***	- 0.017***	(1.25)	- 0.017***	- 0.016***	- 0.016***	(1.14)
Demand	Exports	(0.0057)	(0.0057)	(0.0055)	(0.020)	(0.0058)	(0.010)	(0.010)	(0.020)
Competition	Number of firms	-28.9	-28.4	-9.36	-31.3	-	-	-	-
Competition	Herfindahl (sector-region)	(26.0)	(25.9)	(26.0)	(26.0)	-0.96**	-0.89**	-0.97**	-1.01***
Regulation	Firm Entry+Evit	-	- 0.58	- 0.56	- 0.37	(0.38)	(0.38)	(0.38)	(0.38)
Regulation	Firm Endy (Exit	-	(0.70)	(0.70)	(0.62)	(0.72)	(0.71)	(0.71)	(0.70)
Uncertainty	Volatility of financial credits	110	111	116	115	86.6	-	-	89.6
TT	X7 1 / 11/	(101)	(101)	(102)	(100)	(102)	-	-	(101)
Uncertainty	volatility export prices	-	-	-	-	-	-162^{***}	-	-
Uncertainty	Volatility imports	-	-	-	-	-	-	-602	-
		-	-	-	-	-	-	(433)	-
STI policy	RD&D p.w * .Publications	-0.30	-0.34*	-	-0.30	-0.36*	-0.38**	-0.37	-0.38*
STI policy	RD&D p.w. * Universiduates	(0.20)	(0.20)	-0.17	(0.23)	(0.20)	(0.19)	(0.23)	(0.23)
1 2		-	-	(0.15)	-	-	-	-	-
Spillovers	RD&D p.w * MigUniv	0.40**	0.39**	0.37**	-	0.39**	0.41**	0.41**	0.40**
Spillovers	RD&D n w * Inno sector-region	(0.16)	(0.15)	(0.15)	-0.020	(0.18)	(0.18)	(0.18)	(0.18) -0.0067
Spino (eis	Read p.w. Inno sector region	-	-	-	(0.094)	-	-	-	(0.089)
Demand	RD&D p.w.* Expo	0.00018 (0.00047	0.00017 (0.00048)	0.00015 (0.00049)	0.000029 (0.00047)	0.00013 (0.00045)	0.00027 (0.00048)	0.00027 (0.00050)	0.000079 (0.00044)
Competition	RD&D p.w *NumberFirms) 0.36*	0.38*	0.39*	0.26	-	-	-	-
Competition	RD&D n w * Herfindahl	(0.20)	(0.20)	(0.20)	(0.21)	-	- 0.0064	- 0.0069	-
competition	NDed p.w Hernitean	-	-	-	-	(0.0002)	(0.017)	(0.018)	(0.0072)
Regulation	RD&D p.w. * Firm Entry+Exit	-	0.076	0.069	0.075	0.060	0.062	0.059	0.055
Uncertainty	RD&D p.w.*Volat finan. credit	-31.0***	(0.070) -31.0***	(0.070) -29.2**	(0.070) -31.6***	(0.071) -31.9***	(0.071)	(0.072)	(0.069) -31.6***
Uncertainty	RD&D n w * Volat avec prices	(11.6)	(11.6)	(11.9)	(11.4)	(11.5)	-	-	(11.3)
oncentality	p.w. voiat expo prices	-	-	-	-	-	(10.5)	-	-
Uncertainty	RD&D p.w. * Volat impo	-	-	-	-	-	-	-8.06	-
	Employment (sector-province)	-7.3e-06 (9.7e-06)	-6.5e-06 (9.8e-06)	-4.5e-06 (9.3e-06)	-6.0e-06 (9.9e-06)	-	-	-	-
	Constant	10.5*** (0.49)	10.4*** (0.53)	10.7*** (0.51)	10.6*** (0.50)	10.4*** (0.46)	10.4*** (0.46)	10.4*** (0.46)	10.5*** (0.47)

Table A.3. Results for Equation [2], Including Simultaneously One Indicator for Each Complementary Factor Identified in Figure 5^a

Observati	ons 9,246	9,246	9,246	9,246	9,246	9,246	9,246	9,246
R-squared	0.393	0.393	0.393	0.393	0.394	0.394	0.393	0.395
Number o	f ID 3,489	3,489	3,489	3,489	3,489	3,489	3,489	3,489
AIC	1977.5	5 1977.8	1988	1979.4	1963.4	1969.1	1973.8	1957.5
BIC	2098.7	2113.3	2123.5	2115	2091.8	2097.4	2102.2	2100.1

-

Note: Each specification alternate proxies for complementary factors; columns headings show on what factor such change occurred. p.w. = per worker; R&D +D = research design and development; STI = science, technology, and innovation. *** p<0.01, ** p<0.05, ** p<0.1



Figure B.1. Graphical representation of the 2nd Instrumental Variable (Change in Government Regulation)

Note: ENDEI = Employment and Innovation Dynamics National Survey; RD&D = research design and development.

Figure B.2. Variation of Contextual Factors among Regions

a. Average household income

b. Proportion of people with a university degree



Figure B.3. Variation of Contextual Factors among Sectors



Figure B.4. Correlation between Contextual Factors and Sectoral RD&D Returns



a. Export volatility

b. Immigrants with a university degree



Appendix C. Measurement of Physical Capital Stock

Our strategy to estimate firms' capital stock for the period 2014–16 is described by the following equation:

$$K_{it} = \frac{Total \ Energy \ Costs_{sector-size}}{Total \ Wage \ Costs_{sector-size}} * Wage \ Costs_{it} + \frac{\Delta Energy \ Costs_{2010-12 \ Sector}}{Investment \ in \ machinery_{2010-11 \ Sector}} * Investment \ in \ machinery \ for \ Innov_{it}}$$
[3]

The first term of the sum aims at measuring initial capital stocks at the beginning of each year, while the second is an estimate of each firm's capital formation during that year. The first factor in each term of equation [3] was calculated using information from ENDEI 1 due to lack of such data in the second wave of the survey.

The first term is the average proportion of energy consumption in relation to labor costs (a kind of K/L), for each sector and size group, multiplied by each firm's wage costs. Therefore, if we assume that the relationship between energy costs and wage costs within each sector-size group is stable through time, this multiplication estimates the energy costs of firm i at time t explained by *existing machinery*.

The second term estimates the average energy use of new equipment (which we assume is equal to the investment in machinery) by sector using information from ENDEI 1, which is then multiplied by each firm's investment in new machinery for innovation.³³ Therefore this second multiplication estimates the energy consumption of the firm due to *new machinery*. Both terms of the sum are expressed in units of energy expenses, which makes the addition possible.

Comparing capital intensity across sectors using energy consumption as a proxy gives reasonable and consistent results with data from both ENDEI 1 and the estimated indicator calculated with equation [3] for ENDEI 2. The most capital-intensive sectors were pharmaceutical, others (which includes cars and petroleum), basic metals, and non-metallic minerals.

Table C.1 presents some descriptive statistics for key variables included in equation [1]. Table C.2 presents their correlation matrix. Labor productivity increases with firm size, as do our proxies for labor, capital, and knowledge factors. Coefficients of variation are larger for RD&D than for other factors. Our estimation of capital stock is not particularly dispersed. Capital stock correlates more tightly with value added per worker than knowledge and labor factors—although none of the correlation coefficients appears to be particularly high, but all of them are significant.

³³ In the cases where wage costs were not reported, we imputed the value of wage costs using the estimated average of this variable for the firm's sector-size group.

Firm size		VA per worker (in 10 thousands)	RD&D per worker (proxy for knowledge stock) (in thousands)	Estimated energy consumption per worker (proxy for capital stock) (in thousands)	Labor
Small	Mean	49,49	3,09	18,03	15,70
	Std dev	69,12	16,90	28,90	5,43
	Var coef	1,40	5,47	1,60	0,35
Medium	Mean	60,62	4,86	27,38	47,08
	Std dev	76,49	19,33	28,79	20,80
	Var coef	1,26	3,98	1,05	0,44
Large	Mean	82,27	6,30	46,69	448,94
	Std dev	127,19	19,29	50,15	668,11
	Var coef	1,55	3,06	1,07	1,49
Total	Mean	60,74	4,44	27,35	120,07
	Std dev	88,33	16,40	32,39	353,80
	Var coef	1,45	3,69	1,18	2,95

Table C.1. Descriptive Statistics of Variables in Equation [1]

Note: RD&D = research design and development; VA = value added.

Table C.2. Pearson Correlation Coefficients for Variables Included in Equation [1]

	VA per worker	RD&D per worker (proxy knowledge stock)	Estimated energy consumption per worker (proxy capital stock)	Labor
VA per worker	1			
RD&D per worker (proxy knowledge stock)	0.122	1		
Estimated energy consumption per worker (proxy capital stock)	0.274	0.114	1	
Labor	0.141	0.05	0.274	1

Note: RD&D = research design and development; VA = value added.

Appendix D. Considering Firms' Innovative Capacity by Accounting for Strategy and Managerial Skills

We considered whether firms' specific attitudes on such matters as strategy and managerial skills, which proxy firms' innovative capacity, affect the impact that contextual factors have on RD&D returns. For this, we include variables constructed with answers from Sections 2 and 7 of ENDEI 2, which give information on the firm's attitude toward innovation, its position in the market, and managerial strategy. Interacting these variables with the interaction of RD&D and the contextual variables allows us to see if this interaction is significant for specific types of firms. Given that the firm's susceptibility to the various contextual variables depends on different attitudes that shape its innovative capacity, we chose different specific attitudes that we considered relevant to interact with each contextual factor.

Table D.1. Proxies of Firms' Innovative Capacity to Benefit from Contextual Factors, Built from
Firms' Strategies, Attitudes, and Managerial Skills, by Contextual Factor of Figure 5

Contextual factor interacted	Proxy of <i>innovative capacity</i> to take advantage (or hedge against) contextual factor		
	Name	Explanation	
STI Policy and	STI and	For "Publications," we use a dummy equal to 1 if the firm relies on academic-specific	
Spillovers	SPIL	information sources (Q 7 answers 10, 11, 14, 15). For "Innovation per sector-region," we use a dummy equal to 1 if the firm relies on other firms for information (Question 7 answers 7, 8, 9, 13, 14). These variables are intended to represent the intensity of the firm's relationship with external information sources. Firms with a stronger relationship will probably benefit more from the supply of scientific knowledge and inter-sectoral spillovers.	
Demand	DEM	A variable that indicates whether a firm declares that one of the main motives to innovate were factors related to market demand (Q 7.2, answers 6, 7, 8).	
Competition	COMP	A dummy that is equal to 1 if the firm considers itself a technological leader in the market. These firms are expected to be positively affected by competition, as a more competitive environment will push them to increase their innovation activity (Q 2.2, answer 1).	
Regulation	REGU	A dummy indicating whether the firm mentioned satisfying regulations and rules as a main motivation for innovation (Q 7.2, answer 11).	
Uncertainty	VOLAT	A variable equal to 1 if the firms mentioned macroeconomic uncertainty as an obstacle to innovation (Q 7.3, answer 10).	

To introduce these managerial skills, we followed a procedure similar to the one used to select the contextual variables. First, we estimated separate regressions for each group of contextual factors using as a baseline our estimation of column 8 in table A.3. Results can be seen in table D.2.

In column 1 of table D.3, we join the results for all the groups. We include micro-level interactions only when they add new (relevant) information to the results, which is the case of the first four columns of table D.2:³⁴

³⁴ We considered results to be relevant when the inclusion of the firm-specific characteristic allows us to observe heterogenous RD&D returns across different groups of firms. We do not consider the interaction with our regulation variable relevant because it was not statistically significant. Interaction with spillovers and uncertainty measures are not relevant either, given that they are significant for both groups of firms, and the coefficients are not statistically different.

Table D.2. Milero Interactions: Regressions by Groups of Contextual Variables						
	(1)	(2)	(3)	(4)	(6)	(7)
VARIABLES	STI pol	Spillovers	Demand	Compet	Regul	Uncert
	0.30***	0 30444	0 20444	0 20444	0 20444	0 20444
Capital per worker	0.30^{***}	0.30^{***}	0.30^{+++}	0.30^{+++}	0.50^{+++}	0.50*** (0.022)
PD&D nor worker	(0.032) 0.012**	(0.032)	(0.052)	(0.052)	(0.052)	(0.032)
KD&D per worker	(0.013^{++})	-0.0090	(0.016^{11})	(0.0048)	(0.0055)	(0.012^{111})
Labor	-0 27***	-0.26***	-0.26***	-0 26***	-0.26***	-0.26***
Labor	(0.045)	(0.045)	(0.045)	(0.045)	(0.045)	(0.045)
Year = 2015	0.19***	0.18***	0.19***	0.18***	0.18***	0.18***
	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)
Year =2016	0.33***	0.34***	0.35***	0.35***	0.35***	0.35***
	(0.027)	(0.024)	(0.023)	(0.023)	(0.024)	(0.023)
Missing dummy	0.11**	0.10**	0.097**	0.082*	0.089*	0.092*
	(0.047)	(0.045)	(0.048)	(0.047)	(0.047)	(0.048)
STI=0 * Publications	7.009					
	(8.98)					
STI=1 * Publications	45.48***					
	(10.24)					
STI=0* Publications * RD&D p.w	-0.084					
	(0.269)					
SII=1* Publications * RD&D p.w	-0.60/***					
PD&D = w * STI - 1	(0.231)					
RD & D p.w *SII = I	(0.0039)					
SPIL=0 *MigUniv	(0.007)	2 79				
size v hingoint		(4.44)				
SPIL=1 *MigUniv		-0.394				
6		(2.99)				
SPIL=0 * Inno_sector_region		-0.526				
C		(1.85)				
SPIL=1 * Inno_sector_region		6.65***				
		(1.08)				
SPIL=0 *MigUniv * RD&D p.w		1.70**				
		(0.77)				
SPIL=1 *MigUniv* RD&D p.w		0.34**				
		(0.162)				
RD&D p.w * SPIL=1		0.0296				
		(0.0203)				
SFIL=0 "KD&D p.w. " Inno_sector_region		-0.343				
SPII = $1 \times RD \otimes D$ n w \times Inno sector region		(0.333)				
STIL I ND&D p.w. IIII0_SCOULTERION		(0.166)				
DEM=0* Expo		(0.100)	0.021***			
DELL V EAPO			(0.0061)			
DEM=1* Expo			0.0070			
1			(0.0078)			
DEM=0* RD&D p.w. * Expo			-0.00034			
			(0.00043)			
DEM=1* RD&D p.w. * Expo			0.00088			
			(0.00060)			
DEM=1* RD&D p.w			-0.0097			
			(0.0074)			
COMP=0* Herfindahl				-0.97**		
				(0.41)		
COMP=1* Hertindahl				-1.07		
COMD-0* DD & D * U £ 111				(1.03)		
COMP=0" KD&D p.w. " Herlindahl				0.034^{**}		
				(0.014)		

Tabl D 2 M Int .**+**i D ic h fC ntovtual Variabl

Table C.2 continued

COMP=1* RD&D p.w. * Herfindahl				-0.089*		
				(0.050)		
COMP=1* RD&D p.w				0.0069		
DECU-0* Eine Enter Erit				(0.0088)	0.22	
REGU=0* FirmEniryExit					0.23	
					(0.00)	
REGU=1* FirmEntryExit					1.50	
					(1.04)	
REGU=0* RD&D p.w. * FirmEntryExit					(0.025)	
					(0.077)	
REGU=1* RD&D p.w. * FirmEntryExit					0.11/	
					(0.157)	
REGU=1* RD&D p.w					-0.022	
					(0.023)	46.0
VOLAT=0 ⁺ volatility of financial credits						40.9
						(110)
VOLAT=1* volatility of financial credits						121
						(130)
VOLAT=0* KD&D p.w * volat of financial credits						-38.4^{++++}
						(14.1)
VOLAT=1* KD&D p.w * volat of financial credits						-28.4°
						(10.3)
VOLAI-I' KD&D p.w						-0.0043
Constant	10 2***	10 6***	10 6***	10 0***	10 6***	(0.0032)
Constant	(0, 42)	(0.20)	(0, 27)	10.8^{-11}	(0, 20)	(0, 28)
	(0.42)	(0.39)	(0.37)	(0.39)	(0.39)	(0.38)
Observations	9,246	9,246	9,246	9,246	9,246	9,246
R-squared	0.392	0.392	0.391	0.392	0.390	0.391
Number of ID	3,489	3,489	3,489	3,489	3,489	3,489

Note: COMP = competition; DEM = demand; p. w. = per worker; RD& D = research design and development; SPIL = spillover; STI = science, technology, and innovation; VOLAT = volatility. Clustered standard errors in parentheses. *** p<0.01, ** p<0.05, * p<0.1

Table D.S. Micro Interactions: Joining	ALL VALIAUICS
VARIABLES	(2) Joining
	everytning
Capital per worker	0.29***
	(0.032)
RD&D per worker	0.0034
Labor	(0.015)
Labor	(0.045)
Year = 2015	0.19***
	(0.015)
Year = 2016	0.32***
Missing dummy	(0.028)
Wissing duminy	(0.047)
STI=0* Publications	-0.21
	(9.41)
STI=1* Publications	31.6***
SPII =0* Inno sector region	0.92
	(1.95)
SPIL=1* Inno_sector_region	5.00***
	(1.13)
Univ Migrations	2.48
DEM=0* Expo	(3.13) 0.027 ***
	(0.0072)
DEM=1* Expo	0.0082
	(0.0083)
COMP=0* Hernnaani	-0.94** (0.39)
COMP=1* Herfindahl	-1.09
	(1.07)
Firm Entry+Exit	0.67
Credit valat by sector	(0.70)
credit volat. by sector	(103)
STI=0* RD&D p.w. * Publications	0.065
	(0.30)
STI=1* RD&D p.w. * Publications	-0.65**
SPIL =* RD&D n w * Inno sector region	-0.34
	(0.31)
SPIL=1 *RD&D p.w. * Inno_sector_region	-0.14
	(0.19)
RD&D p.w. * MigUniv	0.32**
DEM=0* RD&D p.w. * Expo	-0.00051
	(0.00040)
DEM=1* RD&D p.w. * Expo	
COMB-0* DD&D n w * Harfindahl	(0.00056)
Com - KD&D p.w. Intrinuani	(0.011)
COMP=1* RD&D p.w. * Herfindahl	-0.091*
-	(0.047)
RD&D p.w. * FirmEntryExit	0.014
RD&D n.w. * Volat credit sect	(U.U68) _ 31 1 ***
icoup p.m. volat_cicuit_scci	-31.1

Table D.3. Micro Interactions: Joining All Variables

Table C.3 continued

	(11.3)
STI=1* RD&D p.w	0.0079
	(0.0077)
SPIL=1 *RD&D p.w	0.0027
	(0.0095)
DEM=1* RD&D p.w	-0.0084
	(0.0071)
COMP=1* RD&D p.w	0.0048
	(0.0087)
Constant	10.5***
	(0.47)
Observations	9,246
R-squared	0.398
Number of ID (sample size)	3,489

Note: COMP = competition; DEM = demand; p. w. = per worker; RD& D = research design and development; SPIL = spillover; STI = science, technology, and innovation; VOLAT = volatility. Clustered standard errors in parentheses. *** p<0.01, ** p<0.05, * p<0.1

ISIC	
Code	
(Rev 3)	Sector
15	Food
1511	Meat industry
1520	Dairy products
1552	Wine and other beverages
17	Textile products
18	Wearing apparel
19	Leather
20	Wood
21	Paper
22	Publishing, printing and reproduction of recorded media
24	Chemical products
2423	Pharmaceutical products
25	Rubber and plastic products
26	Other non-metallic minerals
27	Basic metals
28	Other metal products
29	Machinery and equipment
299	Tools and machinery in general
2921	Machinery for Agriculture
33	Medical instruments
35	Other transport equipment
36	Furniture
2930	Equipment for domestic use
3012	Electrical material, radio and TV
3420	Trailers and semi-trailers
3430	Car parts
9999	Others, includes tobacco (16), cars (341), oil (23), and recycling (37)

Appendix E. ENDEI 2 Economic Sectors

Note: ENDEI = second round of the Employment and Innovation Dynamics National survey; ISIC (Rev 3) = third revision of the International Standard Industrial Classification of All Economic Activities.

Appendix F. Implications for Research and Data Collection

Making Micro Data Sets More Accessible

Micro databases in Argentina are rarely available for research. Although several waves of innovation surveys have been produced since 1992, micro-data were made publicly available only after the Labor and Employment Ministry and the Science, Technology and Productive Innovation Secretariat produced the Employment and Innovation Dynamics National (ENDEI) survey (2010–12). Public dissemination required an effort from these organizations to guarantee confidentiality. This was an important institutional innovation, as it was the first time that economic information was made available at the micro level.

It would be very useful to follow the same procedures for previous innovation surveys (ENIT, National Survey on Innovation and Technology Behavior) managed by the National Institute of Statistics and Censuses, INDEC. A panel with innovation data could be built beginning with the early 1990s with these data sets. Similar recommendations could be made for industrial surveys. In contrast to other countries in the region, in Argentina there is no access to valuable information that could be used for different research purposes, such as providing empirically based policy recommendations.

Building Panels

Panel databases need to be built. Neither ENIT nor ENDEI was built to be matched over time. In fact, the questionnaires are not the same—not even between ENDEI 1 and ENDEI 2. Data are not matched. The resulting panel, if/when matched, will not be representative either at either the sector or regional level because of differences in sampling procedures for each wave. This complicates the use of panel-data econometric techniques, which are advisable to control for biases originated by time-invariant omitted variables.

Matching Databases

It is also necessary to be able to match different databases such as innovation and industrial surveys and administrative data, such as those produced by Customs, the revenue service (AFIP), and the social insurance agency (ANSES), and offer these matched databases for research.

One particular database that would be useful to match is produced by FONTAR, the main program supporting firm innovation offered by the National Agency for Science and Technology Promotion. FONTAR, like ENDEI, depends on the Secretary of Science, Technology and Productive Innovation. FONTAR includes different instruments, as tax credits, subsidized loans, and matching grants. The FONTAR database has information for firms that have applied to FONTAR since 2008. All applicants are requested to fill out an innovation survey, which makes it possible to construct groups for beneficiaries and non-beneficiaries to be used in impact assessments. Although the data are available, access is restricted and has been allowed only for internal evaluation or when FONTAR funders requested it (see Pereira and Suárez 2018, Arza and Vázquez 2014, 2015).

Providing More and Better Sectoral, Regional, and More Aggregated Information

There is a serious lack of sectoral and regional data in Argentina. For example, there is no consistent information on value added disaggregated by sector-province. The latest data available are for 2004. In addition, there is no sufficiently disaggregated information on sectoral value added at the national level over time (at the ISIC 3-digit level). There is very little information about public policy supporting industrial sectors, and even less at the

regional/provincial levels. For example, there are no data on the public budget organized by sectoral activities over time that could be used to assess the impact of public budget expenditures on firms' productivity.

Finally, there is very little information about institutional quality in Argentina (for instance measures of corruption, or "doing business" surveys with enough disaggregation across sectors and regions).

Improving Formats

Certain information is available but only in a very unfriendly format for use in research. For example, international trade data by province are accessible, but must be downloaded in a very fragmented fashion. Similarly, for export/import tariffs or nontariff measures, some information is available, but is presented in an unsuitable way for analysis. This information can be accessed only by entering the product code in a webpage, while no databases are available that summarize the export tariffs for a set of products. Finding data on historic trade tariffs is also extremely complicated.

In addition, it would be good to merge different databases on science and research. For instance, information about agreements signed at liaison offices at universities or National Scientific and Technical Research Council (CONICET) is available only in an unfriendly format. Combining that information with bibliometric information, could be used to build information on knowledge networks across disciplines and geographic locations. This could yield useful insights on regional/sectoral competitiveness and help measure the impact of public support to research activities in the country.

Recommendations Regarding ENDEI

Argentina lacks a measure of capital stock at either the firm or sectoral level. Because panels are not built, it is not possible to use methods of permanent inventory (that is, using firms' investment in physical assets over time). It is therefore particularly important that innovation surveys include variables that could be used as proxies for capital stock, such as energy consumption or investment in physical assets. In ENDEI 2, both measures presented measurement errors and therefore could not be used.

For comparisons across countries, it would be good to follow international guides, such as the Oslo Manuals. In the case of ENDEI, some questions do follow those guidelines (such as innovation expenditures) but others do not (such as barriers to innovation).

Regarding qualitative information, we recommend using Likert scales for each category. This would allow researchers to assess each category independently within a question, instead of asking respondents to rank order (or to choose a number of relevant) different options in a list, as is done in ENDEI 2 in questions 2.1 and 2.2.

One question included in the OSLO manual that used to be part of ENIT questionnaire asks about the importance of different research fields for firms' innovation activities, using the Likert scale. This question is useful to assess to what extent public support to scientific fields matches firms' demand of scientific knowledge. It is not included in ENDEI questionnaires. It might worth reassessing its relevance for future surveys.

Finally, it would be very useful to administer innovation surveys for firms in the service sector, given the importance this sector has for national value added and employment.